Smaller Learning Communities 2006-07 Evaluation Report Inglewood Unified School District Morningside High School Grant# V215L042250

February 2008

Research conducted by:

Michael Butler Mercy Sanchez

Public Works, Inc. 90 N. Daisy Avenue Pasadena, CA 91107 (626) 564-9890 (626) 564-0657 fax

Table of Contents

I.	Introduction Background to the SLC Approach Complementary Reforms to Support Smaller Learning Commu Implementation Issues for Smaller Learning Communities Reform Context in Inglewood Overview of SLC Implementation Public <i>Works</i> , Inc. Evaluation and Report Report Organization	1 nities
II.	Evaluation Methodology Evaluation Approach Literature Review Surveys Site Visit Quantitative Data	13
III.	Quantitative Analysis Student Demographic Characteristics Student Achievement Results Other School Performance Indicators	17
IV.	Qualitative Analysis Description of Morningside High School Vision, Leadership, and Management Teaching and Learning Teams Rigorous, Relevant Curriculum and Instruction Inclusive Programs and Instructional Practices (SLC Structure) Accountability and Continuous Program Improvement School/District Support of SLCs Personalization Parent & Community Engagement	23
V.	Conclusions and Recommendations Summary of Student and School Outcomes Summary of SLC Implementation	37
VI.	Appendices	
	Appendix A: Literature Review Appendix B: Staff and Student Surveys	

Appendix C: SLC Evaluation Checklist

Part I—Introduction

With the leadership of the Gates Foundation in creating a national agenda to fund high school reform and research, public support through the federal Smaller Learning Community (SLC) grants, and consensus on the need to address the persistent problem of high school dropouts and lackluster student performance nationwide, school districts across the nation are transforming large comprehensive high schools into smaller, more manageable units of 200-500 students. Simultaneously, autonomous small high schools (typically new start-up schools or charters) have been developed to provide a more personalized high school experience.

SLC reforms combine with the push for accountability of the standards-based reforms of the 1990s and the No Child Left Behind Act (NCLB). Under the lens of the so-called "New 3R's," SLC reform strategies are intended to match academic achievement (*Rigor*) with curricular approaches that bring meaning and application to students (*Relevance*) along with enhanced personal connections to adults and other students (*Relationships*). As such, SLC reform involves changes that offer what many say is the opportunity for badly needed secondary school improvement—providing what is often lacking in high school education and the possibility for curricular change, meaningful collaboration, and systemic student support.

This report provides evaluation results from 2006-07 for Morningside High School in the Inglewood Unified School District (IUSD). Morningside received a US Department of Education Smaller Learning Communities (SLC) implementation grant in 2004-05.¹ IUSD hired Public *Works*, Inc., a non-profit headquartered in Pasadena, California, to conduct a third-party evaluation of the SLC initiative.

About the US Department of Education Grants

Since 2000, the U.S. Department of Education's SLC grant program has provided planning and implementation grants to high schools with 1,000 or more students in order to plan, implement or expand SLCs. The grants support a range of strategies including creating schools within schools with varying degrees of autonomy,² restructuring the school day to allow for cohort scheduling and more consistent student-adult interactions, and formal adult mentoring and advisory programs.³ Implementation of these structural changes share the goals of a more personalized high school experience for students in smaller schools within schools and to improve student achievement and performance.

Continued under the Bush Administration's No Child Left Behind (NCLB) Act, the program provides three-year implementation grants ranging from \$250,000 to \$550,000 per school. In total, the U.S. Department of Education has awarded over \$734,177,166

¹ The U.S. Department of Education has awarded funds on an annual basis. Cohort III schools are schools who first received implementation funding in the 2003-04 school year. Cohort IV schools received funding in 2004-05. Cohort V schools received implementation funding in 2005-06. Cohort VI schools received funding in 2006-07.

² School-within-a-school refers to an autonomous school that, while it may be in its own building or in a building with another school, is organizationally, fiscally, and instructionally independent.

³ Advisory systems place students under the guidance and care of a teacher or administrator for their entire school experience on a regular (daily or weekly) basis.

through 2006 to schools across the nation, with funding for this program growing each year.

Background to the SLC Approach

High School Student Performance

In the late 1990's, after years of reform focused on implementing standards-based accountability systems which tended to yield improved student outcomes at the elementary level, questions about the stubborn lack of progress among high school students came to the forefront as the new frontier of education reform. Both performance on international assessments and national measures of student achievement indicated the need for dramatic improvement.

In 2003, US students placed 28th in mathematics and 29th in problem solving out of 40 participating countries with sufficient data on the Organisation for Economic Cooperation and Development (OECD) Programme for International Student Assessment (PISA). Further, from 1992 to 2002, the National Assessment of Educational Progress (NAEP) indicated that 60 percent or more of 12th graders performed below the Proficient level (Klekotka, 2005).

The achievement gap continued to be large with African-American and Hispanic students at the end of high school having reading levels equivalent to White eighth-graders (Phi Delta Kappa International, Topics & Trends, Volume 5, Issue 4). Other data suggested that even college-going high school students were unprepared to succeed in college. For instance 25% of freshmen at four-year institutions and 50% of freshmen at two-year colleges did not return for the second year (Phi Delta Kappa International, Topics & Trends, Volume 5, Issue 1).

The persistent and high dropout rate across the nation also began to receive more attention, especially as researchers pinpointed the problems existing in so-called "dropout factories" characteristic of many urban school districts. As the No Child Left Behind Act and state accountability strategies such as exit exams have raised the profile of the number of students who don't complete high school, a key study by Robert Balfanz at the Center for Social Organization of Schools based at Johns Hopkins University identified approximately 2,000 schools in 15 states (one of which is California) that account for 80 percent of high school dropouts located primarily in urban areas, the South, and the Southwest (Balfanz, 2004 and Samuels, 2007).

The 21st Century Take on High School Reform

In 2005, following the National Education Summit on High Schools, the National Governors Association identified an *Action Agenda for Improving America's High Schools* that called on state leaders to: (1) make all students proficient and prepared, (2) redesign the American High School, (3) give high schools the excellent teachers and principals they need, (4) hold high schools and colleges accountable for student success, and (5) streamline and improve education governance.

The actions of the nation's governors followed many years of commission reports, conferences, and research identifying the anonymity, apathy and alienation so prevalent among our nation's youth combined with the overriding consensus that it was driven in large part by the very structure of high school education embodied in large, comprehensive high schools. Launched in 2000, the Gates Foundation five-year high school initiative provided over a billion dollars in funding on a range of fronts—at the individual school level to break up large schools or start new schools, for researchers and policymakers to learn more about effective practices, and most recently, to build capacity at the district level to sustain widespread change.

While high school reform has been characterized by "dozens of actors and innumerable initiatives," reformers are "focusing primarily on five strategies—improving school climate, strengthening curriculum and instruction, raising graduation requirements, helping freshmen get up to speed academically, and preventing students from dropping out" (Toch, 2007, p. 434).

Lessons Learned About the Impact of School Size

Practitioners and policymakers have debated the appropriate size for high schools from at least the mid-20th century when population growth and funding practices resulted in large high schools becoming the norm. Ted Sizer of the Coalition of Essential Schools (organized in 1984) and Deborah Meier (known for her work with Central Park East in New York City in the late 1980's and early 1990's) were among the more vocal and renowned advocates for small, personalized learning environments for high school students. In turn, private foundation funding from the Gates Foundation beginning in 2000 and earlier Annenburg Foundation grants to reform urban schools favored the movement toward small schools or smaller subunits within the larger campus.

Beyond improving academic achievement, research suggested that small schools built a more positive and productive educational environment conducive to student learning. A sense of community constructed through student self-selection, as well as increased staff interest in students, led to greater feelings of belonging and more investment in making the school a quality place to learn. Classroom discipline problems, disruptions, and assaults were found to be less common in small schools, due to an increased sense of community and genuine investment in the school and learning (Cotton, 2001).

Based on these reviews of research and other information from high school students themselves, attention was placed on school size as the "lever" for improving high school student outcomes. However, in their review of the research related to small school size, authors Lee, Ready, and Welner report that "not all small-school news is good" and that "a bit of caution may be in order" (p. 7). They found issues related to privacy in which the reputations of students' siblings or parents preceded them and that small schools often attempted to replicate the more comprehensive curriculum of larger high schools with faculty teaching out of their specialties. The lesson for those attempting to break up large high schools perform. "Much of the enthusiasm for small schools focus on those small schools that *want* to be small, often have selective entrance criteria, and are staffed by innovative faculty and attended by committed students (Lee, 2002, p8)."

Common Approaches to Implementing SLCs

Under the US Department of Education's SLC grant program, implementation grants are provided to high schools with 1,000 or more students in order to implement and expand SLCs. The grants support a range of structures (i.e., reorganization of student placement and staff assignments) and strategies (i.e., techniques and measures to provide interdisciplinary, personalized instruction and guidance to students) including creating schools-within-schools, career academies, restructuring the school day, formal adult mentoring and advisory programs all with the goal to create a more personalized high school experience for students and to improve student achievement and performance (see Table 1 for a summary of common approaches to SLCs).

Table 1: Structures and Strategies	for Small L	earning C	ommunities
---	-------------	-----------	------------

Small	The term "small school" or "school-within-a-school" refers to an autonomous school that,
Schools	while it may be in its own building or in a building with another school, is organizationally,
and	fiscally, and instructionally independent (Small Schools Project, 2001a). Teachers and
Schools-	students are self-selected. The school has its own leader, school-day schedule and classroom
within-	space. Small schools, like other small learning community models, can have a focus, or theme,
Schools	be identified as an "alternative" school, or have a number of other labels attached. Regardless,
	small schools operate autonomously.
Academies	Under the academy model, high schools organize the curricula and education program for a
	subset of students (usually ranging from 200-400 students) around one or more themes,
	typically career or occupationally related. Under the model, a small group of students is
	grouped with a team of teachers responsible for creating interdisciplinary and personalized
	curriculum across career and academic content. Students stay with this team of teachers
	typically for grades 10-12. In addition, career academies partner with employers,
	postsecondary institutions and other community groups to infuse the curriculum and
	educational experience of students with one-to-one mentors, internships, service learning and
	other extracurricular support.
Magnet	Magnet schools, usually with a core focus such as mathematics and science, performing arts or
Schools	humanities, typically draw students from an entire district and have often been used as a
	strategy for racial desegregation of urban school districts. Although magnets are "choice"
	programs open to all, the admission processes are often complicated and include factors such
	as timing of application, race/ethnicity, preferences for existing siblings, transportation
	considerations, teacher recommendations and grades. Magnet students often benefit from
	additional fiscal and personnel resources including a core group of faculty that primarily teach
	within the Magnet and additional individual support through a Magnet director and/or
TT	specially assigned counselor.
Houses	A nouse contains classrooms for teachers of core subjects who function as a team to instruct a
	small group of students (ranging from 100-500) (Sammon, 2000). In some models, students
	in their house. Some schools have used the house model as a way to help freshmen transition
	into the larger high school by offering a separate house for sub sets of the entering freshmen
	late the ranger high school by one mig a separate house for sub-sets of the entering itestiment
	Often, houses can contain a sequence of career related and (or academic courses that lead
	toward graduation (Cotton 2001) Houses are often an alternative ontion for groups aiming
	to produce the same positive student outcomes as small schools, but do not quite have the
	intention funding or resources available to achieve a completely autonomous small school
Other	Comprehensive high schools are devising additional strategies for breaking up the learning
"Small"	experiences of students so that they can form more significant attachments to adults and their
Strategies	neers. Examples of these strategies include:
ollategies	• Advanced courses for high achieving students
	Nerror and a state in the state of the
	• Newcomer schools for inningrant students entering a school system for the first time
	• Modifications to the high school schedule (for example, block scheduling)
	• Ninth-grade house plans similar to houses but involving only the ninth grade
	• Advisory systems in which students are placed under the guidance and care of a teacher or
	administrator for their entire school experience (essentially a personal academic and social
	guidance counselor)

Source: Public Works, Inc.

Complementary Reforms to Support Smaller Learning Communities

As comprehensive high schools break up into smaller units and new schools are started, what is being learned is that size is no guarantee for success. Schools that have experienced the most success have implemented complementary reforms that bring about improvements for student outcomes.

College Prep Curriculum for All

An increase in the rigor of high school courses and adopting a curriculum that supports students as they transition out of high school into college is no longer viewed as at odds with a relevant and supportive environment that encourages students with the least preparation to stay in school. In fact, evaluations of SLC efforts have concluded that the freshmen year is a pivotal year that must address both the need for freshmen with poor academic skills to catch up and to offer them rigorous courses that supports credit attainment and on time graduation (Toch, 2007).

Since 2001, 11 states and LAUSD, the second largest school district in the nation, required students to complete a full college-prep course sequence. In addition, 22 states currently require graduation exams. Many feared that these increases in graduation requirements would result in higher dropout rates. In addition, there was fear that these initiatives requiring more academic coursework runs counter to the notion of relevance and personalized learning.

However, emerging research indicates that may not necessarily be the case and that the combination of rigorous coursework with relevance is supportive of students graduating. For example, one study from Johns Hopkins University found that "enrollment in career-technical education is positively associated with higher graduation rates, but *only* when the tech courses are taken along with more challenging academic courses (Toch, 2007, p. 435)." On the other hand, an evaluation of efforts to raise graduation requirements in Chicago noted that simply calling courses college-prep was not sufficient and that the courses needed to be taught by capable teachers that can provide a challenging curriculum and motivation for students to complete the material (Toch, 2007).

Professional Learning Communities and Distributed Leadership

Another complementary reform to SLCs is to support professional collaboration and distributed leadership among professionals in the new, smaller sub-units. In schools that move beyond structure and discussions of "architecture" as put by Tom Vander Ark, former executive director of the Gates Foundation education initiatives, the development of professional learning communities offers a real opportunity for making instructional change the focus of reforms. According to Richard DuFour, a national expert on the implementation of this kind of reform, professional learning communities focus on three "big ideas": (1) shifting from a focus on teaching to a focus on learning, (2) creating structures that promote a collaborative culture, and (3) an orientation on judging effectiveness based on results (DuFour, 2004).

In the context of SLCs, professional development to support improved pedagogical methods can be delivered within SLC teams, it is also important to complement this professional development within the content areas of teachers departments or specialties (Quint 2006). With more collaboration and targeted professional development, faculty and staff in SLCs and small schools work together to improve curriculum quality. This enables teachers in these settings to teach across content areas and spend more time personalizing curriculum and lessons to address the needs of individual students.

Personalized and differentiated instruction offers teachers more flexibility and more options in teaching students based on what works, which includes considerations for learning styles, socio-cultural influences and possible learning disabilities (US Dept. of Education, 1999). This increased tailoring of education to individual needs contributes to the narrowing of the achievement gap, and at the same time reduces the effects of ethnic minority and poverty, by harnessing group effort and focusing it upon helping all students in the specific areas they need the most (Howley, Strange, and Bickel, 2000).

9th Grade Support Systems

More school districts are focusing on 9th graders because students who fail to earn sufficient credits to matriculate to 10th grade are much more likely to dropout. The *Talent Development* high school model from Johns Hopkins implemented first in Philadelphia and in other districts across the nation focused on providing 9th graders with accelerated "catch-up" courses in reading and math. *Talent Development* high schools offer a double dose of math and English for an entire year (90 minutes each). "During the first semester, they take classes designed to give them the academic and study skills necessary to handle college-prep courses later on; during the second semester, teachers follow the district's regular curricula for English and algebra, supplemented with special materials developed by Johns Hopkins University (Toch, 2007, p. 436)." Students taking this sequence outperformed their peers in comparison schools and even students who started with higher-than-average achievement benefited.

In its evaluation of *First Things First* (a 9th to 12th grade model of theme-based SLCs implemented in Kansas City, Kansas) and *Talent Development* high schools (that incorporate a 9th grade Success Academy with career academies in the 10th to 12th grades), MDRC found that both structures played a positive role in increasing attendance and reducing dropout rates (Quint, 2006). However, the evaluation cautioned that simply increasing the amount of time in English and math classes for freshmen does not result in higher student achievement—*Talent Development* also implements special curricula to maximize learning during the extra time.

Implementation Issues for Smaller Learning Communities

While many high school reformers were entering uncharted territory as the SLC movement took hold, evaluation results and lessons learned are beginning to surface that may help to keep reform on track. Evaluation results funded by the Gates Foundation of its own high school reform initiative, findings from the MDRC evaluation of three widely implemented models, and an evaluation of New York City's New Century High Schools Initiative are just a few examples of recent publications indicating both the promise of and trouble spots to watch out for in the implementation of SLCs. In particular, early SLC implementers quickly learned that though small learning environments often provided the context to make reform possible, the break up into smaller units was only the beginning, not the end of the process.

Conversions vs. Start-ups

Schools, especially in urban districts, have taken a variety of approaches to restructuring high schools including spinning off new schools from closed or reconstituted high schools,

as charters run by other organizations, or conversions of larger schools into smaller subunits with varying degrees of autonomy over decision-making and fiscal responsibility. One of the largest infusions of support for these changes has been the Gates Foundation National School District and Network Grants Program, which also funded an evaluation by the American Institutes for Research (AIR) and SRI International.

Early findings from the evaluation indicated that after the first year of operation, new small high schools had already made great strides in establishing deeper and more supportive student-teacher relationships both academically and personally. However, these environments required a large amount of work to put in place, more than the teachers had first anticipated. Further, the work of establishing a new school was more complicated and time-consuming leading to significant shortfalls of the resources necessary to implement all of the components needed to meet the challenging student populations they had been successful in recruiting. Facilities suitable to these new small schools were difficult to come by and the multiple roles of instructional leaders, personal advisors, and participants in distributed leadership challenged these teachers (AIR/SRI, April 2003).

The evaluation's examination of large school conversions also found that conversions of existing schools take longer than first envisioned with planning encompassing a two-year process. Further, conversion high schools had more difficulty instituting the type of structures for personalization that emerged in new small schools after the one start-up year. Teacher commitment to SLC change in conversions was also more tenuous due, in part, to the fact that SLC planning teams tended to involve a small proportion of teachers at the school (AIR/SRI, April 2003).

Impact of SLCs on Student Achievement

In the most recent round of evaluations of high school conversions and new start-up schools, the impact of SLCs on student achievement is mixed. While many have made progress in a key reform area-improved school climate, there is less conclusive evidence of the impact on student achievement. For instance, the MDRC summary of its evaluations of Career Academies, First Things First, and the Talent Development model found improvements in eleventh-grade math and reading tests in Talent Development schools for students where the interventions had been in place the longest but no effect on achievement within the Career Academies they studied (Quint, 2006). The evaluations of Gates-funded new and converted high schools found some improvements in reading and language arts especially in high schools that had implemented the Foundation's Attributes of High-Performing Schools to a higher degree.⁴ However, their study found poor rigor in mathematics assignments at new and redesigned high schools (AIR/SRI, 2005b). Despite these mixed results related to specific academic content areas and SLCs, the MDRC study of Career Academies found reduced dropout rates, improved attendance, and increased likelihood of on-time graduation among Career Academy students, especially those most at risk off dropping out (Kemple, 2000).

⁴ Gates Foundation Attributes of High-Performing Schools include (1) Common Focus, (2) High Expectations, (3) Personalization, (4) Respect and Responsibility, (5) Time to Collaborate, (6) Performance-Based, and (7) Technology as a Tool (AIR/SRI, 2005).

Autonomy

The issue of autonomy in SLCs goes to the heart of the reform in the breakup of large impersonal and bureaucratic comprehensive high schools. In the context of SLCs, autonomy can have a variety of definitions or approaches. For instance, SLC faculty may have autonomy over various aspects of organizing curriculum and instruction such as scheduling, staffing classes, and the like but little decision-making authority over core components of school organization such as budgeting and hiring decisions. Other aspects of autonomy include procedures for recruiting and selecting students, student conduct, and SLC safety.

The variation in levels of autonomy also presents one of the largest stumbling blocks in implementing the types of learning environments most connected to student success—those that allow for collaboration among adults and personalization for students. As high schools go through the conversion process, school-wide planning often takes three-years or more delaying discussions by SLC teams or schools-within-schools about the central questions of instructional improvement and just what is meant by personalization. In addition, to avoid "community unrest," issues "revolving around ability-grouping, advanced-placement opportunities, band, school spirit, or athletics may take precedence over strong efforts to improve instruction and enhance personalization (Fink and Silverman, 2007)."

Size

While there is no consensus on the "perfect" size for a high school or an SLC, a large-scale quantitative study using nationally representative and longitudinal data explored the ideal size of a high school based on student learning. Using data from 10,000 students in 800 public and private schools in the US, achievement gains in mathematics and reading over the course of high school were found in schools of between 600 and 900 students (a middle-sized high school). However, maintaining an even smaller school size was a more important factor for schools enrolling high proportions of disadvantaged students (Lee, 2002). SLC conversion schools vary greatly in the numbers of students per SLC, which is often dependent on the overall size of the school and the number of SLCs the faculty deems is feasible to implement. For most of the SLCs in high school conversion schools a range of 200 to 400 students per SLC is feasible, particularly in urban settings.

Tracking

Tracking students by their perceived ability is a long-standing practice prevalent in American high schools that has been the subject of deep controversy especially related to the persistent achievement gap for low-income and minority students. While there are both opponents to and advocates for ability-based tracking, researchers are finding that grouping students in SLCs can either serve to dismantle or reinforce low, medium, and high-ability tracks. "What research exists on schools-within-schools suggested that secondary schools that engage in this reform improve their social environments. However, early indications also suggest that the reform may increase internal stratification inside high schools, especially if unrestrained choice is the means used for students to be matched to sub-units (Lee, 2002, p. 34)." In an article describing the "multiple pathways" approach embedded in many SLC reforms, authors Jeannie Oakes and Marisa Saunders describe how important it is to implement programs that consciously allow students to select programs based on their interests rather than being "selected or directed" based on their past achievement, where they are assumed to be going after high school, or their perceptions of the level of difficulty of the courses in a given SLC (2007).

Managing the Master Schedule

Implementing a master schedule that works for all SLCs in a converted high school is one of the biggest challenges to success. Scheduling classes to insure "purity" of teachers and students within the same SLC has been a major challenge to school administrators especially for students in the upper grades who may want to take electives offered by other communities (Quint 2006). Building in more autonomy and a separate identity for each SLC, reducing the number of student and teacher "cross-overs" between SLCs, and allowing for flexibility in the master schedule (i.e., not maintaining a common bell schedule) are all strategies for managing the master schedule in converted high schools. In addition, reducing the number of small, specialized programs may also contribute to SLC purity.

Research on the use of various block scheduling (e.g., 4X4 blocks, alternating A/B days) has not yielded a consensus on the impact of these types of schedules on student achievement. In a comparison of a traditional schedule to a 4X4 block schedule, there were no differences in academic achievement, teacher satisfaction with the schedule, or the use of instructional strategies. However, other research has found that block schedules may result in fewer discipline problems and failures, less time spent on classroom administration, and the opportunity for students to earn more credits with the 4X4 block schedule, a real benefit for students in need of credit recovery (i.e., those who failed academic courses) and/or (Phi Delta Kappa International, Topics & Trends, November 2006, Volume 6, Issue 4).

In *Talent Development* schools, double-blocked schedules were found to be especially useful for freshmen because it allows students to earn more credits per year (i.e., it has a built in safety net for students who fail core academic courses and need to repeat these courses) than other types of scheduling. Traditional scheduling allows for students to attempt fewer courses. Semester-long, intensive "catch-up" courses allow ninth-grade students to have additional support in reading and mathematics, key to staying in school and graduating (Quint, 2006).

Physical Space

A study conducted by the National Center for Education Statistics reported that 14 percent of US public schools are overcrowded and eight percent are severely overcrowded. Moreover, schools enrolling mostly minority students are more likely to be overcrowded than schools with less than half minority enrollment (Lee, 2002). Year-round schedules and multiple tracks are commons strategies for addressing these over crowded schools. Given this context, especially in urban areas, for high schools converting to SLCs, creating space that supports autonomy can be an overwhelming challenge. For instance, locating teachers by SLC may not be possible given the facility's configuration. The traditional organization of most high schools into departments (e.g., English, Math, Science) is also usually reflected in the layout of buildings making it difficult to co-locate a team of teachers from multiple disciplines. This is further complicated in over-crowded schools where teachers must sometimes move from classroom to classroom and where students attend on different year-round tracks.

Reform Context in Inglewood

Located in the southwest portion of Los Angeles County, the Inglewood Unified School District (IUSD) served 15,945 students in grades K-12 in 2006-07. Since 2003-04, IUSD has experienced a 13% decline in enrollment (i.e., fewer students enrolling in elementary grades due to demographic and socioeconomic trends in Southern California). Like other districts in California, secondary schools have had a more difficult time achieving federal accountability targets. None of IUSD's elementary schools have been identified as Program Improvement (PI). By contrast, all but one of the secondary schools are now PI schools.

Morningside High School (the high school receiving the U.S. Department of Education SLC grant) has also been a Program Improvement school since 1998-99. Based on failure to meet State accountability goals 2001-2004, Morningside entered the School Assistance Intervention Team (SAIT) audit process in Fall 2004. Morningside The school became subject to a Corrective Action Plan monitored by the Los Angeles County Office of Education (LACOE). Many of the key corrective actions that the school is implementing supported the goals and objectives of SLCs during the time of the USDE grant including:

- Developing and monitoring a system for implementing a standards-based intervention program based on State Board of Education approved materials.
- Identifying and using multiple measures for ongoing monitoring of student placement in 9th and 10th grade intervention courses.
- > Developing a master schedule that guarantees equity and access for all students.
- Providing teacher teams with time to develop, share, and provide feedback on standards-aligned instructional units.
- Ensuring that regularly scheduled faculty, department and professional development meetings are focused on the review of assessment data and instructional strategies to address the gaps in student achievement.
- Developing and implementing a system and procedure for articulation with feeder middle schools around student achievement data.

Overview of SLC Implementation

Morningside received a three-year SLC implementation grant from USDE beginning in 2004-05. No SLCs existed at Morningside prior to receipt of the grant and 2004-05 essentially functioned as a planning year as no new SLCs were established. In 2005-06 (Year 2), the focus of SLC activities was on the implementation of a 9th grade house structure for all freshmen, which included a designated area on campus. In addition, Morningside piloted a 9th grade advisory (16 advisories, structured as 25:1 student: teacher ratio). There were plans for developing academies for grades 10-12 in 2006-2007 (Year 3), however no progress was made on this front. Instead, the school implemented a 10th grade house in 2006-2007 and an advisory program school-wide.

Public Works, Inc. Evaluation and Report

As required by the US Department of Education, districts receiving SLC Implementation Grants are required to hire a third-party evaluator. The evaluation conducted by Public *Works*, Inc. encompasses both quantitative (student and school outcomes) and qualitative (progress in implementing SLCs as outlined in the grant) analytic approaches.

In order to frame the current evaluation, Public *Works*, Inc. worked with IUSD to develop a set of categories to be used in data collection and to organize the analysis. The categories employed by the evaluation reflect both IUSD's application for SLC funding and researchbased components found to be critical for early implementers of SLCs including:

- 1. Vision, Leadership & Management
- 2. Teaching and Learning Teams
- 3. Rigorous, Relevant Curriculum & Instruction
- 4. Inclusive Programs and Instructional Practices (SLC Structure)
- 5. Accountability and Continuous Program Improvement
- 6. School/District Support for SLCs
- 7. Personalization
- 8. Parent and Community Engagement

Report Organization

Following this introduction, Part II provides a more detailed methodology of the evaluation. Part III provides quantitative data on student and school outcomes for Morningside through the third year of grant implementation. Part IV contains an analysis of the qualitative data regarding program implementation, rating Morningside in the eight categories listed above. Part V concludes the report with recommendations. Appendices include the bibliography, staff survey results, student survey results, and the site implementation checklist used to evaluate SLC implementation.

Part II—Evaluation Methodology

Evaluation Approach

The evaluation of IUSD SLC Cohort IV grant encompasses two dimensions: (1) a quantitative dimension measuring the impact of the grant on student achievement and (2) a qualitative dimension measuring progress with regard to program implementation. The evaluation is an annual process over the three-year grant cycle. Four primary data collection methodologies have been used for the analysis contained in this report:

- A review of the literature related to SLC implementation;
- Staff and student surveys;
- Site visit to the grantee high school; and,
- Compilation of student achievement and school performance data.

Literature Review

The review of literature conducted for this evaluation examined several dimensions of the implementation of SLCs including: the rationale and context for high school reform, a summary of the bodies of research supporting SLCs as a reform strategy, a typology of strategies to implement SLCs and lessons learned from early implementers. Public *Works*, Inc. prepared an extensive bibliography for the literature review, which is included as **Appendix A**.

Surveys

Public *Works*, Inc. developed two surveys of key stakeholders for this evaluation, one for school staff and one for students. The school was provided with the results of the surveys. This was the third administration of the staff and student surveys. The staff and student survey results are contained in **Appendix B**.

Staff Survey

The staff survey was developed to ask all school staff about their knowledge and involvement in the SLC initiative at their school. The survey information provides information about the percentage of school staff self-reporting that they are currently involved in planning or assigned to an SLC and opinions about various aspects of implementation at their school. Teachers, counselors and administrators were the primary respondents to the survey though all members of the staff participating in the school's faculty meeting were invited to complete a survey.

Staff surveys were administered to staff at Morningside in May 2007. In order to calculate a response rate, Public *Works*, Inc. used the CBEDS reported number of certificated staff to estimate the number of staff at the school. In order to ensure a high response rate, the surveys were administered at faculty meetings where all staff was present. The following table displays the response rate for Morningside based on the number of completed surveys. Overall, Public *Works*, Inc. achieved a 47% response rate in 2007.

	# of certificated staff	# of completed surveys	Response rate
2004-05	97	57	59%
2005-06	110	49	45%
2006-07	93	44	47%

Table 2: Staff Survey Response Rates

The analysis of the staff survey included overall frequencies and area means. In addition, Public *Works*, Inc. examined cross-tabulations of results by SLC participation, the number of years of teaching and by subject area. A chi-square test was performed on the cross-tabulations in order to determine statistical significance at the .05 level.

Student Survey

In order to provide a baseline assessment of student opinions and experiences in high school, students were surveyed with regard to their expectations for learning, classroom instruction, counseling and guidance, and personalization. Students were also asked to identify if they currently participate in a small learning community and their participation in activities such as after-school programs, college courses, internships and the like. The survey concluded with demographic questions including grade, sex, race-ethnicity, highest-level math class and plans after graduation in order to track student responses to small learning community implementation over time.

Public *Works*, Inc. administered surveys to 9th and 12th graders. CBEDS enrollment numbers were used to calculate the number of students enrolled at these two grades. Overall, Public *Works*, Inc. achieved a 58% response rate for freshmen and 58% for seniors.

	9th grade enrollment	# of surveys completed	Response rate	12th grade enrollment	# of surveys completed	Response Rate
2004-05	454	333	73%	294	109	37%
2005-06	430	353	82%	294	120	41%
2006-07	365	211	58%	287	168	58%

Table 3: Student Survey Response Rates

The analysis of student surveys included overall frequencies and area means as well as responses summarized for the school. Cross tabulations were prepared by gender, ethnicity, and perception of school. A chi-square test was performed on the cross-tabulations in order to determine statistical significance at the .05 level.

Site Visit

In order to provide qualitative information regarding the implementation of SLCs at the school level, Public *Works*, Inc. conducted a site visit at Morningside. The site visit was scheduled by Public *Works*, Inc. staff and coordinated by the school's assistant principal with responsibility over the SLC grant. The site visit consisted primarily of interviews and focus groups of administrators, staff and students at the school. In order to speak with a range of school stakeholders, Public *Works*, Inc. met with the following:

[•] SLC grant site administrator

- District SLC grant coordinator
- Principal and other administrators
- Teachers (SLC and Non-SLC)
- Counselors
- Students grades 9-12 (SLC and Non-SLC)

To prepare for the site visit, Public *Works*, Inc. reviewed demographic and achievement data for the school. Public *Works*, Inc. held a training for the site visit team prior to the site visit, which included a review of the overall goals for the site visit, background information and a review of the protocols developed specifically for the site visit.

In order to analyze and summarize the data collected during the site visit, Public *Works*, Inc. used an implementation checklist prepared specifically for this evaluation. Survey and site visit information was summarized in the checklists completed. The checklist is included in **Appendix C**.

The Site Visit Checklist provides a means to measure an overall average rating of the status of implementation for individual areas within the initiative. The eight areas rated on the checklist for the SLC grant included:

- 1. Vision, Leadership & Management
- 2. Teaching and Learning Teams
- 3. Rigorous, Relevant Curriculum & Instruction
- 4. Inclusive Programs and Instructional Practices (SLC Structure)
- 5. Accountability and Continuous Program Improvement
- 6. School/District Support for SLCs
- 7. Personalization
- 8. Parent and Community Engagement

The following rating scale was used to provide a gauge of the level of implementation of individual components of small learning communities based on survey results and site visits. The scale incorporates a rubric of both effectiveness of implementation and coverage of the school community, which is broadly defined as students, teachers, staff, administrators, parents and community partners as appropriate to the particular strategy.

Checklist Rating Scale

1=No Evidence of Implementation. Strategies have not been developed; few or no school community members involved and/or impacted; planning to take place in the future.

2=Planning for Implementation. Strategies are in the planning stages; some or a few school community members are involved in planning; few or no school community members impacted.

3=Early Implementation. Strategies are moving beyond planning to implementation; school community members are being recruited for implementation and participation; some school community members impacted.

4=Developmental Implementation. Strategies have moved into implementation; implementation at the early developmental stages; impact on school community is growing.

5=Solid Implementation. Strategies are in solid implementation stage; impact on participants is evident but continues to be fine-tuned.

6=Full Implementation. Strategies are fully implemented; 100% of target school community is participating and impact is positive.

Quantitative Data

In order to assess the impact of SLCs on student achievement and school performance, this report presents outcome data for the past three years (2004-05, 2005-06, and 2006-07) in relation to 2003-04 (baseline prior to receipt of the USDE SLC grant). The student achievement and school performance indicators included in this report include:

- California Standards Test (CST), English Language Arts and Mathematics;
- California High School Exit Exam (CAHSEE), English Language Arts and Mathematics;
- Graduation Rate; and
- College Entrance Examinations.

Part III—Quantitative Analysis

This section of the report presents student achievement and school performance data for Morningside High School, a Cohort IV school receiving U.S. Department of Education funding to implement Smaller Learning Communities (SLCs). This report presents student achievement data for the three-year period of SLC grant implementation (2004-05, 2005-06, and 2006-07).

Descriptive analyses were performed on multiple achievement indicators including:

- California Standards Test (CST), English Language Arts and Mathematics;
- California High School Exit Exam (CAHSEE), English Language Arts and Mathematics;
- Graduation Rate; and
- College Entrance Examinations.

Student Demographic Characteristics

In 2006-07, the majority of students at Morningside were Hispanic/Latino (60%) followed by African-American (38%) students (see Table 4). Similarly, a majority (78%) of the student population qualified for Free/Reduced Price meals. Less than one-third (29%) of the student body was identified as English Learners. Special Education students comprised 10% of student enrollment. In the last three years, there has been very little change in the school's demographic composition other than an increase of students eligible for Free/Reduced Meals (+40%).⁵

	2004	-2005	2005	-2006	2006	-2007
	Number	% of	Number	% of	Number	% of
	(N)	Enrollment	(N)	Enrollment	(N)	Enrollment
Ethnicity						
African American	579	38.6%	592	38.6%	524	38%
Hispanic/Latino	894	59.6%	917	59.7%	814	60%
Other	26	1.8%	26	1.7%	25	1.8%
Special Populations						
English Learners	452	30.2%	387	25.2%	391	28.6
Special Education	137	9.1%	140	9.1%	139	10%
Free/Reduced Meals	561	37.9%	889	58.3%	1.060	78%

Table 4: Summary of Student Demographics at Morningside High School

Source: California Department of Education

Student Achievement Results

To analyze student proficiency in English language arts (ELA) and mathematics, this report focuses on the California Standards Test (CST) and the California High School Exit Exam (CAHSEE). The CST is administered every Spring to IUSD students and scored as part of

⁵ The increase in the percentage of students eligible for Free/Reduced Meals is likely due to school policies to identify and secure completed applications from students rather than a change in the type of student attending Morningside.

the State's Standardized Testing and Reporting Program (STAR). The purpose of the CST is to assess students' performance in relation to the California Academic Content Standards. These standards, adopted by the State Board of Education, are grade and content specific and outline what students in California are expected to know and be able to do. Based on their performance, students are assigned one of the following five proficiency levels: Advanced (A), Proficient (P), Basic (B), Below Basic (BB) and Far Below Basic (FBB). A student who performs at or above the Proficient level is considered to have met the State standards.

Beginning in 2005-06, no student will receive a public high school diploma without passing the English/Language Arts and Mathematics portions of CAHSEE. The primary purpose of CAHSEE is to significantly improve achievement in public high schools and to ensure that students graduate with grade level competency in reading, writing, and mathematics. Students begin taking CAHSEE in the 10th grade and have until the 12th grade to pass the exam. High school students must score a 350 or higher in both subject areas to pass CAHSEE.

English/Language Arts

Approximately half of students (grades 9-11) at Morningside scored in the lowest proficiency levels (Far Below Basic or Below Basic) from 2005-2007 on the CST for English/Language Arts (see Table 5). In the 9th-10th grade, which has been the focus of SLC efforts, the proportion of Advanced or Proficient students improved slightly since 2005. Additionally, there was a decline in the proportion of 9th graders scoring in the lowest proficiency levels from 2005 to 2006, however, no change from 2006 to 2007. Put another way, 9th grade students in Below Basic have not moved into the Basic category since 2006. However, there were more 10th grade students scoring in the lowest proficiency levels from 2005 to 2006 (+6%), followed by a decrease of 10% in these low proficiency levels from 2006 to 2007.

	200	05 (N=1	,112)	200	06 (N=1	,000)	20	2007 (N=965)		
English Language Arts	% Advanced or Proficient	% Basic	% Below Basic or Far Below Basic	² % Advanced or Proficient	% Basic	% Below Basic or Far Below Basic	% Advanced or Proficient	% Basic	% Below Basic or Far Below Basic	
Grade 9	20%	31%	49%	20%	36%	44%	21%	35%	44%	
Grade 10 Grade 11	11% 19%	34% 27%	54% 53%	12% 11%	29% 27%	60% 62%	16% 12%	35% 27%	50% 62%	

Table 5: CST ELA Proficiency Levels by Grade Level

Source: California Department of Education

Data from 2005-2007 show that more than half of the 10th grade students successfully passed the English/Language Arts section of the CAHSEE on their first try (see Table 6). This percentage has decreased 6% from 2005 to 2006, then rebounded with an 8% increase from 2006 to 2007. There were increases in the pass rates between 2006 and 2007 for all subgroups with the exception of English Learner students. Most notably, male students experienced a 13% increase from 2006 to 2007. It is important to note that all of the 10th graders in 2007 were enrolled in an SLC at Morningside, which focused on increasing CAHSEE pass rates.

		2005			2006		2007			
	Number	#	%	Number	#	%	Number	#	%	
Groups	Tested	Passing	Passing	Tested	Passing	Passing	Tested	Passing	Passing	
TOTAL	367	208	57%	408	208	51%	339	201	59%	
Male	205	112	55%	203	80	39%	174	91	52%	
Female	162	96	59%	205	128	62%	164	109	66%	
African American	138	69	50%	170	84	49%	119	66	55%	
Hispanic	222	134	60%	229	119	52%	212	128	60%	
Other Ethnicities	7	*	*	*	*	*	*	*	*	
English Learners	125	49	39%	115	33	29%	97	288	29%	
Economically Disadvantaged	183	111	61%	268	133	50%	280	166	59%	
Special Education	34	1	3%	8	*	*	5	N/A	N/A	

Table 6:	CAHSEE	ELA	Pass	Rates	(10^{th})	grade	only)
----------	--------	-----	------	-------	--------------------	-------	-------

Source: California Department of Education *Not numerically significant

Mathematics

From 2005-2007, the vast majority (80% or more) of students at Morningside scored in the lowest proficiency levels (Far Below Basic or Below Basic) on the CST for Mathematics (Algebra, Geometry, and Algebra II). As shown in Table 7, the scores of 9th graders were slightly higher than those of 10th or 11th graders but the overall level of Mathematics achievement at Morningside is very low. Moreover, Mathematics achievement on the CST has declined across all grade levels since 2005.

Table 7:	CST Mathem	atics Proficien	cy Levels by	Grade Level
			J J	

		2006				2007						
Mathematics	Number Tested	% r Advanced or Proficient	% Basi	% Below Basic or Far Below Basic	Number Tested	% r Advanced or Proficient	% Basic	% Below Basic or Far Below Basic	Number Tested	% Advanced or Proficient	% Basic	% Below Basic or Far Below Basic
Grade 9	322	5%	17%	78%	396	2%	15%	82%	311	2%	11%	87%
Grade 10	342	1%	7%	93%	334	1%	6%	92%	268	0%	2%	97%
Grade 11	460	2%	6%	93%	271	1%	3%	96%	168	1%	0%	99%

Source: California Department of Education

Nonetheless, there has been a dramatic increase in the percentage of 10th grade students at Morningside that successfully pass the Mathematics section of the CAHSEE on their first try (see Table 8). This year, slightly more than half of 10th graders passed the CAHSEE, with the exception for English Learners (29%). Increases occurred among all subgroups and most markedly for Male (18%) and African American (16%) students. It is important to note that the Mathematics CAHSEE primarily tests grade 7 standards (70% of the exam).

	2005				2006		2007			
	Number	#	%	Number	#	%	Number	#	%	
Groups	Tested	Passing	Passing	Tested	Passing	Passing	Tested	Passing	Passing	
TOTAL	363	158	44%	383	165	43%	336	187	56%	
Male	202	84	42%	183	72	39%	173	99	57%	
Female	161	74	46%	200	93	47%	163	88	54%	
African American	137	57	42%	147	58	39%	118	65	55%	
Hispanic	219	98	45%	227	102	45%	209	118	56%	
Other Ethnicities	7	*	*	*	*	*	*	*	*	
English Learners	123	35	28%	115	30%	26%	95	27	28%	
Economically Disadvantaged	181	81	45%	254	109	43%	272	150	55%	
Special Education	33	3	9%	1	*	*	4	*	*	

Table 8: CAHSEE Mathematics	Proficiency	(10^{th})	grade only)
------------------------------------	-------------	--------------------	-------------

Source: California Department of Education

*Not numerically significant

Other Performance Indicators

Graduation Rate

As shown in Table 9 below, the graduation rate at Morningside High improved 19% from 2003-04 to 2005-06 (from 80% to 99%). The graduation rate⁶ was determined by dividing the number of 12th grade graduates by the sum of number of dropouts over a four-year period and number of 12th grade graduates. Morningside virtually eliminated 9th grade dropouts (from 35 to 1) and eliminated the number of 10th grade dropouts (from 21 to 0). Data on 2006-07 is not currently available from CDE.

Table 9: Graduation Rate

	Dropouts Grade 9 through Grade 12	Grade 12 Graduates	Graduation Rate
2003-04	57	233	80.3%
2004-05	22	283	92.8%
2005-06	3	227	98.7%

Source: California Department of Education

Nonetheless, a follow-up survey conducted on recent graduates indicates that a majority of former students were enrolled in a post-secondary program (63% in 2007). Moreover, of the students that were not enrolled in school, 72% planned on enrolling in the coming year.

⁶ The calculation is outlined in *California's Consolidated State Application for No Child Left Behind* and is based on the National Center for Education Statistics (NCES) definition for graduation rate.

IUSD Small Learning Communities Evaluation Report, 2006-2007

Lable	10: Self-	Reported Data on school enrollment	
	Are you currently enrolled I in any school?	f you are not currently enrolled in school or co	llege, do you plan to enroll in the next year or two?
2004-	Yes No	Yes	No
05	64% 36%	n.a.	n.a.
06	70% 30%	71%	29%
07	63% 37%	72%	28%

T.1.1. 10. C.16 D

Most (52%) of students enrolled in a post-secondary program reported attending a twoyear community college with 41% saying that they were attending a four-year college or university.

	In what type o	of school are you er	nrolled? (mark one)
	2004-05	2005-06	2006-07
Four-year college or University	43%	40%	41%
Public Community College	36%	23%	52%
Less than 2-year private or public vocational school	12%	7%	8%
Intend to achieve Master's degree	3%	13%	n.a.
No degree plans	9%	17%	n.a.

Table 11: Self-Reported Data on school-type enrollment⁷

College Entrance Exams Performance

As a prerequisite for admittance, most four-year colleges or universities require students to provide results of either the SAT or the ACT assessment. Both of these tests are tools used by colleges and universities to assess the academic potential of student applicants. Though either test qualifies applying students for admission to most colleges, the major differences are that the SAT tests critical thinking and reasoning and the ACT assesses students' comprehension of academic content. These academic assessment tools are crucial components of a student's college application and provide some indication of eligibility for postsecondary education at institutions of higher education.

As shown in Table 12 below, the average Verbal SAT score among Morningside's students has increased slightly since 2003-04 and remained unchanged in Mathematics. Nonetheless, scores in 2006 were well below the State average of 495 for verbal and 516 for math. The average ACT score at Morningside decreased slightly between 2003-04 and 2004-05. ACT data for 2005-06 was unavailable at the time of this report's writing. It is important to note that none of the 12th graders included in Table 12 would have been involved in a SLC for the three years reported below.

⁷ Percentage of students enrolled in a community college in 2006-07 is artificially inflated

		SAT Scores		ACT	Scores
	Verbal Avg. Score	Math Avg. Score	Number Tested	Average Score	Number Tested
2003-04	391	387	115	16.7	53
2004-05	386	372	113	15.8	29
2005-06	404	387	194	n.a.	n.a.

Table 12: Average SAT and ACT Scores

Source: California Department of Education

PART IV—QUALITATIVE ANALYSIS

This part of the report summarizes the analysis of the qualitative data collected for this evaluation. After a brief sketch of Morningside's approach to SLC implementation, this section focuses on rating Morningside in terms of eight dimensions of SLC implementation:

- 1. Vision, Leadership & Management
- 2. Teaching and Learning Teams
- 3. Rigorous, Relevant Curriculum & Instruction
- 4. Inclusive Programs and Instructional Practices (SLC Structure)
- 5. Accountability and Continuous Program Improvement
- 6. School/District Support for SLCs
- 7. Personalization
- 8. Parent and Community Engagement

Within each of these areas of implementation, Public *Works*, Inc. has included a benchmark, which describes what a fully implemented high school organized around SLCs for all students would look like in each area. Following the benchmark for each area, Public *Works*, Inc. has provided an overall rating based on site visit, survey, and quantitative outcome data.

The rating is based on the following six-point scale: 1—No evidence of implementation; 2—Planning for implementation; 3—Early implementation; 4—Developmental implementation; 5—Solid implementation and 6—Full implementation.

Description of Morningside High School

Morningside High School is a comprehensive high school on a traditional calendar. In 2006-07, the student body was comprised of 1,365 students (60% Hispanic/Latino and 38% African-American). English Learners (EL) comprised 29% of total pupil enrollment, 97% of whom spoke Spanish as their primary language. Morningside also enrolled 10% Special Education students.

No SLCs existed at Morningside prior to receipt of the grant. In 2005-06, Morningside implemented houses for 9th graders. In 2006-07, 10th grade houses were also implemented and the advisory period was expanded to all students 9-12. At the end of three-year grant period, 54% of the student body was enrolled in a SLC (i.e., well short of the federal goal for 100% or wall-to-wall implementation. Staff overwhelmingly agreed that SLC implementation was often eclipsed by other school improvement efforts such as, WASC and SAIT.

Vision, Leadership, and Management

Benchmark: Implementation is characterized by a shared vision created by a group of educators, support staff, administrators, parents, and community who comprise the school learning community who assume responsibility for the learning of every student through a distinctive and focused standards-based curriculum.

	▼(2.0 Ave	erage in 2004-05))		
1	2	3	4	5	6
No evidence of	Planning for	Early	Developmental	Solid	Full
implementation	implementation	implementation	implementation	implementation	implementation
▼(3.0 Average in 2005-06)					
▼(3.0 Average in 2006-07)					

In 2006-2007, Morningside's SLC implementation consisted of 9th-10th grade houses in geographically separate parts of the campus. Four teams each serving approximately 100 9th-10th grade students were implemented so that the group of students in each house stayed together most of the school day. Students in the 9th grade house generally took English, Algebra I, and an elective course together, while those in the 10th were cored in their English, Geometry, and World History courses. Additionally, these students shared an advisory class. Morningside did not make progress in terms of developing theme-based academies to serve 11th-12th grade students (i.e., academy structures that 10th graders would select from). Instead, juniors and seniors were incorporated into a daily advisory period.

Morningside did succeed in involving a wider group of staff in the development of the school's SLC goals/vision in the third year of grant implementation. For instance, department chairs were brought into conversations regarding looping and student advisories, as well as discussions about possibly changing the course sequence in Math. Nonetheless, most staff uninvolved in the 9th-10th grade houses remained uncertain about the direction of SLC implementation and saw very few connections between SLCs and other school improvement efforts.

As shown in Table 13 below, survey responses of staff from Spring 2007 continued to highlight the need for better school-wide communication. For example, the survey revealed that after three years of the grant less than half (44%) of staff felt that the vision for SLC was clear. Indeed, 47% of staff respondents were completely unaware of the school's vision as it pertains to SLCs. Moreover, only 28% said that they have a say in school decisions and one-third (35%) agreed that they receive communication once major decisions are made.

The vision and goals for implementing small learning communities are well understood by staff.	44%
The school has a strong leadership team that guides instruction and the implementation of SLCs.	42%
The results of major school decisions are communicated to all staff.	35%
All staff members have a say in school decisions.	28%
Most staff members at this school trust one another.	24%

Table 13: 2007	Staff Percentions	of Vision &	Management	(% Agreement)	(N=44)
Table 13. 2007	Stan i erceptions		management	(/º Agreement)	(11-77)

In addition, interview with school staff indicated that there is confusion about the school's priorities. Some staff view SAIT and WASC (accreditation visit scheduled for Spring 2007) as more important than implementing the SLC grant. Put another way, Morningside had <u>not</u> come to consensus on how or whether SLCs will be a vehicle for transforming secondary education. Consequently, the vision for a wall-to-wall SLC reform effort never materialized at Morningside.

In terms of the Vision, Leadership, and Management benchmark, Morningside would be well-served to focus on the following:

- Communicating the school-wide SLC vision as well as concrete, short-term goals related to SLC implementation to all staff.
- Clarifying issues that relate to SLC versus departmental priorities.
- Taking active steps to make the development of the master schedule more transparent to staff.
- Showcasing linkages and overlap between SLC implementation, compliance with SAIT corrective actions, and WASC accreditation.

Teaching and Learning Teams

Benchmark: SLC teams have structured opportunities for interdisciplinary teaching and learning in order to collaborate and work with small groups of students. SLC teams share no more than 300 students and team members are assigned to SLCs for at least half of their school day. Common planning time and professional development resources are used to support SLC goals and to reflect on student work and performance and to adjust curriculum and instructional strategies.

▼(1.0 Av	erage in 2004-05)			
1	2	3	4	5	6
No evidence of	Planning for	Early	Developmental	Solid	Full
implementation	implementation	implementation	implementation	implementation	implementation
	▼2.0 (Average is	n 2006-07)			
	$\mathbf{V}(2.0 \text{ Average in})$	n 2006-07)			

In 2006-07, professional development was largely focused on content support by coaches (especially Math/English) for less experienced teachers, pacing, and assessments. School-wide professional development also concentrated on preparing for WASC. As a result, there was less time available for teachers to meet in interdisciplinary SLC teams (i.e., 9th and 10th grade house teachers). For example, staff met once during the second semester in

order to discuss SLC vision as it pertains to advisories.

Survey responses of 9th-10th grade house teachers were significantly less positive in this area of grant implementation (see Table 14). More than one-third (42%) survey respondents disagreed that SLCs were able to meet regularly for planning and even fewer (23%) noted that teachers have a role in designing professional development related to SLC implementation. Consequently, most staff saw no connection between school-wide professional development and SLC implementation efforts. In focus groups, comments included the following: "They [school and district] don't support teaming" and "It's always a learning year and never an action year."

Tuble 14. Otali i olooptions of redoning a Leaning (% Agreement) (n	
SLC team members meet regularly for planning, curriculum, and activities.	38%
SLC topics are a regular feature of school-wide professional development.	33%
There is sufficient time for teachers to support students' academic and personal needs and to help them plan for the future.	32%
Professional development for the SLC initiative is designed by teachers and is specific for our school.	25%
There is sufficient time for teachers to discuss and analyze student work in SLC team meetings.	19%

Table 14: Staff Perceptions of Teachin	g & Learning (% Agreement)	(N=44)
---	----------------------------	--------

Teachers in the 9th and 10th grade houses were especially interested in devoting more professional development and collaboration time to personalization strategies. Indeed, on the survey only 38% felt that there was adequate time for supporting students academic and personal needs and only 32% said that sufficient time was allocated for discussing and analyzing student work.

In terms of the Teaching and Learning Teams benchmark, Morningside would be wellserved to focus on the following:

- Allocating more time for teachers to meet by SLC/house.
- Providing teachers work more time to discuss and analyze student work.
- Involving teacher leaders in the development of the school's professional development calendar.

Rigorous, Relevant Curriculum and Instruction

Benchmark: A standards-based educational program embodies high expectations for every student to achieve grade-level standards and meet high school graduation requirements. Students are expected to meet college entrance requirements and are prepared for postsecondary education and training experiences and the world of work. Curriculum is adapted to individual student learning needs and is organized around topics of interest to students. Multiple forms of assessment provide opportunities to reflect personalized learning.

	▼(2.0 Av	verage in 2004-05	5)		
1	2	3	4	5	6
No evidence of	Planning for	Early	Developmental	Solid	Full
implementation	implementation	implementation	implementation	implementation	implementation
$\mathbf{\nabla}(2.0 \text{ Average in } 2005-06)$					

▼(3.0 Average in 2006-07)

The findings from Morningside's SAIT audit clearly articulated the need for improving alignment of teaching and learning to State content standards, increasing student engagement and teacher expectations for students, providing interventions to students struggling to meet standards, and improving assessment practices.

Against this backdrop, SLC planning concentrated on implementing reforms to improve high school graduation rates and CAHSEE scores. For example, the 9th-10th grade house structures were designed, in part, to enroll the majority of students in parallel intervention courses, effectively "doubling" the time spend on English and/or Mathematics.⁸ Similarly, the implementation of the advisory period was intended to assist students in meeting graduation requirements and preparing for postsecondary education.

Although SAIT, WASC, and SLC have all communicated a common message about increasing academic rigor and scaffolding instruction so that more students are able to access academic content at grade level, there is limited evidence of systemic improvement in student achievement. As indicated in Section III of this report, student achievement remains quite low, with improvements concentrated in English/Language Arts and in CAHSEE preparation.

Despite ongoing low levels of student achievement, the majority of staff were satisfied with the school's instructional program. Staff responses in this area suggest that instructional delivery is rigorous, responsive to student needs, and data-driven.

Table 15: Staff Perceptions of Rigorous, Relevant Curriculum and Instruction(% Agreement) (N=44)

Students understand classroom academic expectations	86%
Curriculum and instruction is organized so that all students are expected to learn and perform at high levels.	77%
Instruction is culturally responsive and accommodates diverse student interests, learning styles, and educational needs.	74%
School-wide instructional decisions usually take into account the needs of English Language Learner (ELL) students.	72%
Examination of disaggregated student data is a regular part of school planning and assessment.	70%
There is a clear, connected and comprehensive model for monitoring student progress.	52%

Student survey responses also paint a moderately positive picture of the school's instructional program (see Table 16). For example, most students reported clear academic expectations and teaching at a high level of rigor. Fewer students agreed that classes were interesting and challenging, or encouragement to take AP or advanced coursework.

⁸ 9th grade students who did not need English or Math remediation participated in a Geography elective.

	9 th Grade (N=333)	12 th Grade (N=109)
I will be prepared to enter college when I am finished with high school.	74%	75%
I can get tutoring and other help if I'm having trouble in school.	68%	71%
My teachers are clear about what they expect from me.	66%	75%
Teachers and administrators encourage me to challenge myself.	62%	73%
Teachers teach academic subject matter at a high level.	62%	67%
My teachers are fair about how they grade me.	59%	69%
My classes are interesting and challenging.	55%	57%
I have the opportunity to do assignments and projects about interesting topics in class.	55%	58%
I have been encouraged to take AP and advanced classes.	33%	57%

A majority (74%) of staff survey respondents agreed that students had opportunities for learning that extended beyond the regular day (e.g., after-school programs, internships, etc.), and half were confident about the process for referring a student to academic intervention. For their part, most students also responded positively when surveyed about access to tutoring. However, only 33% of 9th graders and 21% of 12th graders reported attending after-school programs in 2006-07.

Some of the discrepancy between student/school outcomes and survey data may be attributed to school staffing. As shown in Table 17, Morningside has had a low proportion of fully credentialed teachers relative to district averages and quite lower than State averages in each of the last four years. Similarly, Morningside has a higher proportion of 1st and 2nd year teachers compared to the district in each of the last four years. Combined with the fact that 9th (and 10th) grade courses tended to be staffed by the newest and least experienced teachers, the staffing characteristics of Morningside may explain some of the lack of widespread academic improvement in student outcome measures.

	Fu	lly	Intern	or Pre-	Emerge	ency or	Averag	e Years	# 1 st (or 2 nd
	Creder	ntialed	Int	ern	Wa	iver	Teac	hing	Year T	eachers
	School	IUSD	School	IUSD	School	IUSD	School	IUSD	School	IUSD
2003-04	54%	62%	20%	19%	29%	25%	11.2	10.8	33%	25%
2004-05	56%	65%	25%	27%	23%	18%	8.9	9.0	34%	23%
2005-06	58%	73%	25%	17%	19%	12%	8.5	8.9	41%	26%
2006-07	62%	75%	21%	16%	20%	11%	10.3	9.4	29%	24%
Net Change	8%	13%	1%	-3%	-9%	-14%	-0.9	-1.4	-4%	-1%

 Table 17: Staffing Characteristics, 2003-04 to 2006-07

Source: California Department of Education

In terms of the Rigorous and Relevant Curriculum and Instruction benchmark, Morningside would be well-served to focus on the following:

- Working with the district to ensure that Morningside is fully staffed at the start of each school year with credentialed teachers in core subject areas.
- Continuing professional development and coaching in the areas of academic rigor, relevance, and culturally responsive pedagogy.

• Building upon the SLC structures to include a more focused concentration on pedagogy, personalization, and other strategies for engaging and supporting students.

Inclusive Programs and Instructional Practices (SLC Structure)

Benchmark: Each SLC has a coherent educational program and approach that is known and shared by students, staff, families and community partners. SLC membership is based on students' and teachers' interest and choice to ensure equitable access. SLCs have a unique academic identity and include distinct, heterogeneous group of students and an administrator or teacher leader that leads a cohesive faculty team. SLC teams make decisions related to: curriculum, instruction and assessment; budget, personnel and facilities; master schedule and student programming; and student conduct and issues of community safety. SLCs range in size from 100 to 300 students and students are blocked for at least one half of the school day in an SLC.

▼ (1.0 Av	erage in 2004-05)				
1	2	3	4	5	6	
No evidence of	Planning for	Early	Developmental	Solid	Full	
implementation	implementation	implementation	implementation	implementation	implementation	
▼(3.0 Average in 2005-06)						
▼(3.0 Average in 2006-07)						

During 2006-07, a house model was implemented that included all 9th-10th grade students at Morningside High School. Thus, the current SLC structure is inclusive and equitable. Moreover, efforts at the 9th-10th grade are appropriate in terms of size and the master schedule ensures that 9th-10th grade students stay together most of the school day.

Nonetheless, staff awareness of these important structural changes in place at Morningside is quite low. As shown in Table 18, 23%-51% of staff checked "Don't Know" when asked about the school's structural effort to promote equity and enhance opportunities to learn via SLC reforms. Staff were least informed about parent/student outreach.

Table 18: Staff Perceptions of Inc	lusive Programs &	& Instructional Practices
(N=44)		

	% Agreement	% Don't Know
Most staff at this school are committed to the principle that "all children can learn."	70%	16%
SLCs include heterogeneous groupings of students and are not tracked by student ability.	66%	28%
Admission to SLC is open and inclusive.	62%	23%
SLCs provide information and outreach about their programs to high school students and parents.	44%	30%
SLCs provide information and outreach about their programs to middle school students and parents.	28%	51%

Staff perceptions of the prominence and role of SLCs on campus also show low levels of awareness (see Table 19) despite three years of SLC grant implementation. On average, 26% of staff checked "Don't Know" when asked about the degree of SLC autonomy

and/or identity on campus. Additionally, staff involved in the SLC initiative were least positive regarding the degree of influence and autonomy granted to SLCs. These results suggest the SLCs remain largely under the purview of school leadership and faculty directly involved. Put another way, SLCs are not seen as a reform that exerts school-wide influence.

Survey Item	%	% Don't					
	Agreement	Know					
The school's master schedule supports SLCs.	61%	20%					
SLCs have administrators or teacher directors who lead a cohesive faculty.	45%	17%					
SLCs have distinct physical boundaries.	43%	19%					
SLCs have unique academic identities.	37 %	26%					
SLCs have an educational philosophy that is shared by students, staff, families, and community partners.	31%	29%					
SLCs make decisions related to master schedule and student programming.	31%	28%					
SLCs make decisions related to student conduct and issues of community safety.	21%	26%					
SLCs make decisions regarding budget, personnel, and facilities.	9%	35%					

Table 19: Staff Perceptions of SLC Identity (N=44)

For their part, students reported mixed levels of identification with SLCs. Most equated SLCs with the advisory class (i.e. my SLC is my advisory class), others were familiar with the term SLC but are not aware of the purpose behind it. Among school administrators, one has been given primary responsibility for the SLC grant; other administrators have not been assigned to SLC teams.

In terms of the Inclusive Programs and Instructional Practices benchmark, Morningside would be well-served to focus on the following:

- Increasing student identification within a particular SLC team.
- Highlighting SLC accomplishments and milestones during faculty meetings and school-wide professional development.
- Widening administrative "ownership" of the SLC initiative by assigning more administrators to distinct SLC responsibilities.

Accountability and Continuous Program Improvement

Benchmark: Members of the SLC work together, share expertise, and exercise leadership to ensure that student achievement and personal success is the intended result of all decisions. Internal and external sources of school data are used to make decisions. SLC teams retain primary responsibility, appropriate autonomy and are accountable for making decisions affecting the important aspects of the small learning community.

▼ (1.0 Av	erage in 2004-05)		
1	2	3	4	5
No evidence of implementation	Planning for implementation	Early implementation	Developmental implementation	Solid implementation
		· 2005 0		

^{▼(2.0} Average in 2005-06) ▼(2.0 Average in 2006-07)

6 Full implementation In 2006-07, Morningside benefited from increased availability of formative assessment data. As part of a district-wide secondary Mathematics focus, quarterly assessments of Algebra I data were administered and scored for the first time. In addition, the emphasis under SAIT for more data-driven planning focused attention on diagnosing student needs. For example, the large number of "intensive" intervention students (i.e., those two or more grade levels behind) was used as a rationale for parallel classes (double-blocking) for 9th grade students. Survey data (see Table 15) suggest that 70% of staff agree that the school is regularly using disaggregated student achievement data to guide decision-making.

At the same time, only limited progress was made on SLC autonomy and *school-wide* accountability for SLC success precisely because SLC reforms were eclipsed by SAIT and WASC preparation in the minds of many staff. Staff who thought they would have more autonomy and a direct role in shaping school reform were disappointed. Insufficient time for SLC teams to collaborate also negatively influenced staff morale of those assigned to the 9th and 10th grade houses, while also suggesting to uninvolved faculty that SLC reforms were less of a priority. Survey also suggest that faculty want more time devoted to examination of student work and linking data to pedagogy.

In terms of the Accountability and Continuous Program Improvement benchmark, Morningside would be well-served to focus on the following:

- Showcasing to staff how data has been used to shape school improvement efforts as well as results from existing programs/initiatives.
- Continuing to train teachers to interpret student assessment data and, more importantly, use these data to modify and refine teaching practices.

School/District Support for SLCs

Benchmark: School and district policies and practices support the implementation of SLCs. School-wide and departmental goals, professional development, scheduling and staffing align with and support SLC needs. Members of the SLC teams retain primary responsibility, appropriate autonomy and accountability for decisions related to individual SLCs.

▼(1.0 Av	erage in 2004-05)			
1	2	3	4	5	6
No evidence of implementation	Planning for implementation	Early implementation	Developmental implementation	Solid implementation	Full implementation
	▼(2.0 A ▼(2.0 A	Average in 2005-0 Average in 2006-0	06) 07)		

In 2006-07, Morningside continued to focus on addressing the SAIT corrective action steps. These efforts are focused on improving the quality of instruction through alignment to State standards, using grant funds to provide content coaching to teachers, and building academic intervention programs/options for struggling students.

While more staff now view SLCs as an important mechanism for school-wide instructional reform, SAIT continued to overshadow the SLC initiative as the primary focus of the school. In addition, staff expressed wariness about "overreach" as Morningside attempted to comply with the requirements of SAIT, SLCs, and now WASC (accreditation visit

planned for Spring 2007). In sum, the existence of multiple, and potentially competing priorities tended to convey the impression of an unfocused, over committed school.

One key success for Morningside was the development of a master schedule that embodied the school's reform priorities. In each of the prior three school years, Morningside's SLC effort had been hampered by the lack of a master schedule that was inadequately planned or not publicized to staff. As a result, Morningside experienced a poorly organized opening of the school year. Thanks to proactive planning and greater transparency, Morningside was able to develop a Fall 2007 master schedule that prioritized 9th and 10th grade coring for SLC teams, as well as a daily advisory period. In 2007, more than half (61%) of staff agreed that SLCs are supported within the master schedule.

At the district level, support centered on: 1) assisting Morningside in accessing and using data to guide and inform school improvement efforts through the development of a formative assessment system in Mathematics; 2) appointing a new principal to Morningside; 3) providing ongoing professional development for secondary Math teachers and principals on research-based, standards-aligned teaching and learning; 4) monitoring SLC grant budget expenditures; 5) promoting alternative scheduling to address intervention needs

Unfortunately, district support was not evident in the area of staffing. Historically high levels of staff turnover at Morningside have meant that many of the teachers staffing the 9th-10th grade house were new (2 years or less teaching experience) teachers, many of whom acknowledged a need for additional support with classroom management skills, content/pedagogy, and personalization.

In terms of the School/District Support benchmark, Morningside would be well-served to continue focusing on the following:

- Emphasizing the importance of 9th grade transition with specific interventions for freshman
- Creating a professional development calendar for the school prior to the end of the school year that balances the need for content-oriented training and collaboration within departments with interdisciplinary forums for SLC teams to collaborate around personalization and thematic curricula.
- Highlighting commonalities among and across SAIT, SLCs, and WASC in terms of school structures and strategies for improving student achievement.
- Formulating areas where SLC teams will have some degree of autonomy (e.g., master schedule, staffing, discretionary budgets, assessment, student conduct, etc.)

At the district level, IUSD should focus on the following:

- Regularly attending and facilitating school SLC planning/governance meetings.
- Staffing Morningside with credentialed teachers in core academic subjects (particularly English and Mathematics); IUSD should "audit" the school's master schedule for balance of teacher placement by experience so that more veteran, credentialed teachers are assigned to 9th and 10th grade students.
- Coaching or training administrators on how to adapt their role to SLCs (i.e., devolving more authority to teacher leaders, moving away from strictly functional responsibilities to oversight for an entire SLC).
- Monitoring the effectiveness of embedded academic interventions (i.e., those offered during the school day) for students 2+ grade levels behind).

• Assisting Morningside in the establishment of partnerships with parents, community, employers, postsecondary institutions and others as part of SLC implementation.

Personalization

Benchmark: Each student's educational experience is characterized by sustained and mutually respectful personal relationships with students, faculty and administrators. Students are known by a group of educators who advise/advocate for them and work closely with the student and his or her family over time. The size of the SLC is appropriate to its vision and mission ranging in size from 100 to 300 students.

	▼(2.0 Ave	erage in 2004-05))			
1	2	3	4	5	6	
No evidence of	Planning for	Early	Developmental	Solid	Full	
implementation	implementation	implementation	implementation	implementation	implementation	
▼(2.0 Average in 2005-06)						
	▼(2.0 A	verage in 2006-0	07)			

Personalization is the primary rationale for implementing SLCs at Morningside High School. Through a more personalized educational experience, Morningside aims to increase student engagement, high school graduation rates, and postsecondary eligibility.

More than half of staff survey respondents agreed that students are experiencing personalized instruction (see Table 20). More than half also agreed that students are receiving verbal counseling and a safe learning environment. Staff were least optimistic about personalized counseling and guidance. For example, only 36% agreed that all students have adult advocates⁹ and only 23% agreed that students complete a written educational plan on high school and postsecondary goals.

Table 20:	Staff Perceptions	of Personalization	(N=44))
-----------	--------------------------	--------------------	--------	---

	% Agreement
Students experience a safe learning environment.	72%
Students experience personalized instruction that is based on diverse learning styles and multiple intelligences.	68%
Students have opportunities to work with one or more teachers over multiple years (e.g., looping and student advisories).	68%
Students receive verbal counseling regarding their secondary and postsecondary course plan from teachers and/or counselors.	68%
Students experience personalized instruction that blends academic rigor with projects that reflect students' interests, life experiences, and culture.	65%
All students have an adult advocating for their academic and personal needs.	36%
Students complete a written educational plan that encompasses goals for high school and postsecondary education with teachers and/or counselors.	23%

Student survey respondents were less sanguine about the extent of personalization in place at Morningside. For example, less than half of all students surveyed reported feeling safe while at school and only about half felt connected to a school-wide community (see Table

⁹ Unlike their peers with more experience, the majority of teachers with less than five years of teaching experience responded positively to this survey item in a statistically significant manner.

21). Overall seniors were more positive compared to freshmen. For example, 42% of seniors agreed that they have an adult at the school to go to for support compared to only 32% of freshman.

	9 th Grade (N=211)	12 th Grade (N=168)
I have an adult at this school that I can go to for help with school and for personal support.	62%	67%
I feel that I belong to a school-wide community.	49%	60%
I feel safe when I am at school.	38%	48%
I have worked with a <i>teacher</i> to develop a written educational plan that reflects my needs and interests. ¹⁰	32%	42%
I have worked with a <i>counselor</i> to develop a written educational plan that reflects my needs and interests.	28%	46%
I talk to my teachers or a counselor regularly about my high school education plan.	23%	56%

 Table 21: Student Perceptions of Personalization (% Agreement)

Like staff, students were least positive about personalized counseling and guidance. Both 9^{th} and 12^{th} grade students tended to *disagree* that they had worked with either a counselor or a teacher on a written educational plan. Few students said that they talked with a counselor or teacher regularly about their plans for the future. Moreover, among the seniors who said that they met with a counselor more than three times in 2006-07, most (73%) reported that these meetings were focused on immediate needs (e.g., selecting courses) rather than a post-high school plan.

Student focus groups provided limited evidence of personalization, particularly counseling and guidance. For example, students reiterated their concern that the school appears more focused on high school graduation than helping students plan for life after high school. Students also indicated that they would like the advisory period improved to include more organized discussions around careers (career day, interests, scenarios), college preparation, teen issues/current events, as well as provide a time for students to showcase their interests/talents.

In terms of the Personalization benchmark, Morningside would be well-served to focus on the following:

- Personalizing counseling/guidance in a more systemic way that identifies roles for both teachers and counselors in helping students set goals for high school and beyond, as well as regular monitoring of student progress toward these goals.
- Using data (formative assessments, grades, etc.) to personalize instruction and/or intervention.
- Developing a true "curriculum" or menu of options for the advisory period.
Parent & Community Engagement

Benchmark: All members of the SLC are viewed as critical allies and are included in the school community (i.e., students, teachers, support staff, parents, administrators, business and community partners). Ongoing partnerships are aimed at supporting continuous improvement of student achievement and personal success. Authentic engagement of school partners leads to sustained participation of partners in decision-making and implementation of school efforts.

▼(1.0 Av	erage in 2004-05)					
1	2	3	4	5	6		
No evidence of	Planning for	Early	Developmental	Solid	Full		
implementation	implementation	implementation	implementation	implementation	implementation		
▼ (1.0 A	▼(1.0 Average in 2005-06)						

▼(2.0 Average in 2006-07)

In 2006-07, brochures on the 9th and 10th grade houses were created for parents. In addition, principal met with nearly all parents of in-coming 9th and 10th graders. Staff noted an increased parental presence before/after school around the 9th-10th grade houses as well as increased articulation between parents and advisory teachers. Nonetheless, the overall level of parent involvement at Morningside remains low and plans have not been implemented to make parent/community involvement more of an overt focus. To date there has been little focus on building partnerships with business or community organizations to assist in SLC implementation.

As shown in Table 22, staff members were not very positive about parent/community involvement. Indeed, more than one-third (43%) of the staff survey respondents cited parent involvement as a top area in need for improvement at Morningside.

Table 22: Staff Perceptions of Parent/Community Involvement (N=44)	
	% Agreement
Community partners, employers and businesses are involved in the development of small learning communities.	47%

. ----..

Parents are considered key collaborators and contributing members to the school 44% community. This school encourages partnerships with employers, postsecondary institutions 19% and others necessary to implement small learning communities.

Students were more positive about parent involvement, particularly parent-teacher interactions and parental support of learning at home (see Table 23).

Table 23: Student Perceptions of Parent/Community Involvement (% Agreement)

	<u>, </u>	
	9 th Grade (N=211)	12 th Grade (N=168)
My parents feel comfortable with my teachers if they have questions or need information.	73%	73%

In terms of the Parent/Community Involvement benchmark, Morningside would be wellserved to focus on the following:

• Providing opportunities for parents/community members to participate in school planning and decisions related to SLCs.

• Offering parent meetings/outreach to explain the accomplishments and direction of SLC implementation.

PART V—CONCLUSIONS AND RECOMMENDATIONS

Summary of Student and School Outcomes

On all measures of student achievement and school performance, Morningside High School is significantly underperforming on high stakes measures such as the CST and CAHSEE after three years of the grant. Quantifiable success was most evident in higher CAHSEE pass rates, particularly for male students, and higher graduation rates (i.e., fewer 9th and 10th grade dropouts). Morningside also met 2007 Academic Performance Index (API) targets for the first time in many years. Nonetheless, Morningside continues to have clear challenges in improving student achievement, especially in Mathematics.

Summary of SLC Implementation

During 2006-07, Morningside expanded SLC structures to all 9th and 10th grade students, reaching 54% of student enrollment. Grade level houses were provided with set-aside space on campus and students in the houses were cored in three or more courses. In addition, the advisory period was expanded to include grades 9-12. No progress was made in terms of developing thematic SLC academies in 2006-07 (the third year of a three-year grant). As such, it is unclear whether or how SLC efforts will reach more students. Moreover, it is clear that SLC implementation was hampered throughout the grant period by the existence of multiple competing reform priorities (notably State-mandated SAIT corrective actions and preparation for WASC).

Morningside achieved only planning status in terms of implementing SLC benchmarks on a school-wide basis.

Implementation Area	Implementation Status
Vision, Leadership & Management	Early
Teaching and Learning Teams	Planning
Rigorous, Relevant Curriculum & Instruction	Early
Inclusive Programs and Instructional Practices (SLC Structure)	Early
Accountability and Continuous Program Improvement	Planning
School/District Support for SLCs	Planning
Personalization	Planning
Parent and Community Engagement	Planning

In order to successfully implement SLCs, Morningside must a) develop structures for grades 11-12 so that more staff are involved in SLC and b) involve all staff in delivering personalized instruction that augments academic rigor with differentiated and responsive pedagogy, curricular relevance, and enhanced counseling and guidance tied to postsecondary education. Key recommendations for addressing these and other aspect of SLC implementing in need of additional focus and attention include:

• <u>Vision, Leadership, and Management</u>. There is a need for broader communication of the school-wide vision, including clarity on issues that relate to SLC versus departmental priorities. The vision for SLCs must be delivered so that more staff

understand that the SLC initiative is connected to SAIT corrective actions and WASC insofar as SLC reforms intend to personalize instruction and otherwise engage and support students.

- <u>Teaching and Learning Teams</u>. Professional development should allocate more time for teachers to meet by SLC/house, preferably focused on discussion and analysis of student work and assessment data. In addition, teachers should play a larger role in creating the school's professional development calendar.
- <u>Rigorous and Relevant Curriculum and Instruction</u>. Professional development and coaching continue to be needed in the areas of academic rigor, relevance, and culturally responsive pedagogy.
- <u>Inclusive Programs and Instructional Practices</u>. Morningside should widen administrative "ownership" of the SLC initiative by assigning more administrators to distinct SLC responsibilities. Faculty effort should center on increasing student identification within a particular SLC team.
- <u>Accountability and Continuous Program Improvement</u>. Morningside should continue to train teachers to interpret student assessment data and, more importantly, use these data to modify and refine teaching practices. Communication to staff should continually showcase how data is being used to drive school decisions.
- <u>School/District Support</u>. Top priority should center on highlighting commonalities among and across SAIT, SLCs, and WASC in terms of professional development, school structures and other strategies for improving student achievement. Morningside should also begin to formulate areas where SLC teams will have some degree of autonomy (e.g., master schedule, staffing, discretionary budgets, assessment, student conduct, etc.)
- <u>Personalization</u>. Counseling/guidance should become more systemic by clarifying the roles of both teachers and counselors in helping students set goals for high school and beyond, as well as regular monitoring of student progress toward these goals. In addition, the advisory period needs to become formalized such that there is a common expectation regarding staff and student responsibilities.
- <u>Parent/Community Involvement</u>. Morningside needs to provide more opportunities for parents/community to participate in school planning, decisions, etc. related to SLCs. Morningside should also offer parent meetings/outreach to explain the accomplishments and direction of SLC implementation.

Morningside's SLC implementation efforts have also highlighted a number of areas where district-level actions are necessary including:

• <u>Teacher Recruitment and Retention</u>. IUSD must prioritize teacher recruitment and retention for lower-achieving schools. At Morningside High, the lack of fully credentialed staff in core academic areas as well as the propensity to disproportionately staff 9th (and to a lesser extent 10th) grade courses with newer teachers likely diminished the impact of reform efforts on student achievement.

- <u>Master Schedule Development and Oversight</u>. IUSD should assist schools in terms of designing a master schedule that prioritizes school improvement goals prior to the end of each school year. IUSD should "audit" the school's master schedule for balance of teacher placement by experience so that more veteran, credentialed teachers are assigned to 9th and 10th grade students.
- <u>Professional Development and Support for Administrators</u>. IUSD should coach and train administrators on how to adapt their role to SLCs (i.e., devolving more authority to teacher leaders, moving away from strictly functional responsibilities to oversight for an entire SLC). In addition, administrators need training on implementation and monitoring of academic intervention programs and services for both "intensive" and "strategic" students.
- <u>Professional Development and Support for Teachers</u>. IUSD should continue funding coaching and professional development in the core areas of English/Language Arts and Mathematics in order to help Morningside (and other schools) meet state and federal accountability targets.
- <u>Parent/Community Outreach</u>. District personnel should play a larger role in helping schools establish partnerships with parents, community, employers, and postsecondary institutions.

Appendix A: Bibliography

Allen, Lili, and Cheryl Almeida, Adria Steinberg. (2001). Wall to Wall: implementing Small Learning Communities in Five Boston High Schools. Providence, RI: Northeast and Islands Regional Educational Laboratory, Brown University.

Ancess J. (1998). Urban dreamcatchers: planning and launching new small schools. In M. Fine and J.I Somerville (eds.) Small schools, big imaginations: A creative look at urban public schools (pp.22-35). Chicago: Cross City Campaign for Urban school Reform.

Azcoitia, C. (1995). Report and recommendations on small schools in Chicago. Chicago, IL: The Small Schools Task.

Barker, R. and Gump, P. (1964). *Big school, small school: High School size and student behvier*. Stanford, CA: Stanford University Press.

Beckner, W. (1983). *The case for the smaller school*. Bloomington, Indiana: Phi Delta Kappa Educational Foundation. ED 228 002.

Berlin, B.M. and R.C. Cienkus. (February 1989). Size: The Ultimate Educational Issue? *Education and Urban Society* 21: 228-231.

Bickel, R., and Howley, C. (2000). The influence of scale on school performance: A multi-level extension of the Matthew Principle. *Education and Policy Analysis Archives*, 8(22), Retrieved from: http://epaa.asu.edu/epaa/v8n22.html

Blank, R.K. (1990). Educational effects of magnet high schools. In W. Clune and J. Witte, eds. *Choice and Control in American Education*, (vol.2., pp77-109). New York, NY: Falmer Press.

Brand, Bety (2003). *Rigor and relevance: A new vision for career and technical education, A white paper.* American Youth Policy Forum: Washington, DC.

Bryk, A.S., & Thum, Y.M. (1998). The effects of High School organization on dropping out : An exploratory investigation. *American Educational Research Journal*, 26, 353-383.

Butler, Michael and Jia Wang, Mikala Rahn, Patricia O'Driscoll, Van Villanueva (2002). Evaluation of the UNITE-LA School-to-Career Partnership: PLUS Evaluation of LAUSD Career Academies. Pasadena, CA: Public Works, Inc.

Butler, Michael and Patricia O'Driscoll, Marisela Perez, Jia Wang (2000). School-to-Career Models in Los Angeles: The Academy Approach. Pasadena, CA: Public Works, Inc.

Canady, R.L. and Rettig, M.D. (1995a). Block Scheduling: A Catalyst for Change in High Schools. Princeton, NJ: Eye on Education.

Carnegie Corporation of New York. (1998). *Turning point: Preparing American youth for the 21st century*. The report of the task force on education of young adolescents. New York: Author. (ERIC Document Reproduction Service No. ED 312 322).

Cawelti, G. (1995). High school restructuring: What are critical elements? NASSP Bulletin, 79(569), 1-15.

Conant, J.B. (1959). The American High School today: A first report to interested citizens.(1st.ed) New York: McGraw-Hill.

Conant, J.B. (1967) The comprehensive high school, New York: McGraw-Hill.

Cotton, K. (1996). Affective and social benefits of small-scale schooling. Charleston, WV: ERIC Clearinghouse on rural Education and small school. (Eric document Reproduction service No. ED 410 088).

Cotton, K. (1996). School Size, School Climate, and Student Performance. From the School Improvement Research Series, Close-Up #20. Portland, OR: Northwestern Regional Educational Laboratory.

Cotton, K. (1996). School size, school climate. And student performance. *CLOSE-Up.2*. Portland, OR: Northwest Regional Educational Laboratory (ERIC Document reproduction Service No. ED 397-476).

Cotton, Kathleen (2001). New Small Learning Communities: Findings From Recent Literature. Portland, OR: Northwest Regional Educational Laboratory.

Crain, R.L. & Strauss, J.K. (1986). Are smaller high schools more or less effective? Baltimore: Center for organization of school, Johns Hopkins University.

Creating schools that work: Lessons for reform from successful urban high schools. (November 2003). Jobs for the Future, Boston, MA and The Center for Collaborative Education, Boston, MA.

Cushman, K. (1999). How small schools increase student learning, and what large schools can do about it. *Principal*, 79(2), 20-22.

Darling-Hammond, L., Ancess, J., McGregor, K., & Zuckerman, D. (1995). *The coalition campus schools project: Inching toward systemic change in New York City.* New York: National center for Restructuring Education, Schools, and Teaching, Teachers College, Columbia university.

Duke, D.L., and Perry, C. (1978). Can alternative schools succeed where Benjamin Spock, Spiro Agnew, and B.F. Skinner have failed? *Adolescence*, 13(51), 375-392.

Dynarski, M., and Gleason, P., Rangarajan, A., and Wood, R. (1998). *Impacts of School Restructuring Initiatives. Final Report.* Princeton, NJ: Mathematic Policy Research, Inc.

Edmonds, R. (1986). Characteristics of effective schools. In U. Neisser (Ed.), *The school achievement of minority children: New Perspectives* (pp.93-104). Hillsdale, NJ: Lawrence Eribaum.

Edwards, C.M., Jr. (November 1995). The 4x4 plan. Educational Leadership, 16-19.

Elliot, M.N., Hanser, L.M., and Gilroy, C.L. (1998). Evidence of Positive Student Outcomes in JROTC Career Academies. Washington, DC: RAND. Prepared for the Office of the Secretary of Defense.

English, F.W. (1993). Changing the cosmology of the school schedule. In L.W. Anderson & H. J. Walberg (Eds.), *Timepiece: Extending and enhancing learning time*. (pp.23-29). Reston, VA: National Association of Secondary School Principals.

Fetler, M. (1989). School dropout rates, academic performance, size, and poverty: Correlates of educational reform. *Educational Evaluation and Policy Analysis*, 11 (2), 109-116.

Fine, M. Ed. (1994). Chartering Urban School Reform: Reflections on Public High Schools in the Midst of Change. New York, Teachers College Press.

Fine, Michelle and Janis L. Somerville (1998). Small Schools Big Imaginations: A Creative Look at Urban Public Schools. New York, NY: Teachers College Press.

Foothill Associates (1997). California Partnership Academies: 1995-96 Evaluation report.

Fouts, J. (1994). A School Within a School: Evaluation Results of the First Year of Restructuring Effort. Seattle, WA: Seattle Pacific University, School of Education. (ERIC Document Reproduction Service no. ED 370 195).

Fowler, W. J., Jr. (1995). School size and student outcomes. In H. J. Walberg (Series Ed.) & B. Levin, W. J. Fowler, Jr., and H. J. Walberg (Vol. Eds.), *Advances in education productivity: Vol. 5.* Organizational influences on productivity (pp. 3-25). Greenwich, CT: Jai Press.

Fowler, W.A. (1992). What do we know about school size? What should we know? Paper presented at the Annual Meeting of the American Educational Research Association, San Francisco, CA.

Fowler, W.J. & Herbert J. (1991, Summer). School size, characteristics and outcomes. *Educational Evaluation and Policy Analysis*, 13, pp. 189-202.

Gamoran, A. (1996). Student achievement in public magnet, public comprehensive and private city high schools. *Educational Evaluation and Policy Analysis*, 18(1), 1-18.

Garbarino, J. (1978). The human ecology of school crime: A case for small schools. In E. Wenk & N. Harlow (Eds.), *School crime and disruption: Prevention models*. Washington, DC: National Institute of Education.

Garbarino, J. (1979). Some thoughts on school size and its effects on adolescent development. *Journal of Youth and Adolescence*, 9(1), 19-31.

George, P.S., and Lounsbury, J.H. (2000). *Making Big Schools Feel Small*. Westerville, OH: National Middle School Association.

Gewertz, C. (September 2000) "Gates Foundation Awards \$56 million For Small Schools". *Education Week*, Vol. 20,Number 02. (http://www.educationweek.org/ew/ewstory.cfm?slug=02gates.h20)

Glass, Gene, Ed. (1982). School class size: Research and policy. Beverly Hills, California: Sage Publications. ED 217 111.

Goodlad, J.I. (1984). A Place Called School: Prospects for the Future. New York, NY: McGraw Hill.

Grant, G. (1994, August). Schools where kids are known. Network News & Views.

Green, G., & Stevens, W. (1988). What research says about small schools. Rural Educator, 10 (1), 9-14.

Greenleaf C.L. (1995, April). You feel like you belong: Student perspectives on becoming a community of *learners*. Paper presented at the annual meeting of the American Educational Research Association, San Francisco, CA.

Gregory, T. (1992). Small is too Big: Achieving a Critical Anti-Mass in the High School. Position paper presented for the Hubert H. Humphrey Institute for Public Affairs and the North Central Regional Educational Laboratory. http://www.smallschoolsproject.org/articles/download/smallistobig.pdf

Gregory, T. (April 2001). Fear of Success? Ten Ways Alternative Schools Pull Their Punches. *Phi Delta Kappan*, 82(8), 577-581.

Gregory, T. (December 2001). Breaking up large high schools: Five common (and understandable) errors of execution. ERIC Digest, EDO-RC-01-6.

Grubb, W.N. (1995). Reconstructing urban schools with work-centered education. *Education and Urban* Society, 27(3), 244-259.

Haller, E., Monk, D., & Tien, L. (1993). Small schools and higher-order thinking skills. *Journal of Research in Rural Education*, 9 (2), 66-73.

Harvey, J. and N. Housman (May 2004). Crisis or possibility? Conversations about the American high school. National High School Alliance, Washington, DC.

Heinbuch, S.E. & Wagner, F. (1992). Small schools operating costs: Reversing assumptions about economies of scale. New York, NY: Public Education Association.

Heller, E.J., Monk, D.H., Bear, A.S., Griffith, J. & Moss, P. (1990). School size and program comprehensiveness: Evidence from High school and beyond. *Educational evaluation and policy analysis* 12, (2), 109-120.

High Schools of the Millenium (August 2000). Report of the Workgroup, American Youth Policy Forum.

Hill, D. (2001). Breaking up is hard to do. Teacher Magazine, October, 31-37.

Hill, P. (1990). High schools with character, vii. Rand Corporation study.

Horn, L. and X. Chen (1998). *Toward resiliency: At-risk students who make it to college*. Office of Educational Research and Improvement, US Department of Education, Washington, DC.

Howley, C. B. (1989). Synthesis of the effects of school and district size: What research says about achievement in small schools and school districts. *Journal of Rural and Small Schools*, 4 (1), 2-12.

Howley, C. B. (1994). The academic effectiveness of small scale schooling (an update). ERIC Digest. Charleston, WV: ERIC Clearinghouse on Rural Education and Small Schools. (ERIC Document Reproduction Service No. ED 389 503)

Howley, C. B., & Bickel, R. (2000). Results of a four-state study: Smaller schools reduce harmful impact of poverty on student achievement. Washington, DC: Rural School and Community Trust.

Howley, C., Strange, M., and Bickel, R. (2000). Research about school size and school performance in impoverished communities. (ERIC Digest, EDO RC-00-10).

Huang, G. and C. B. Howley. (Winter 1993). Mitigating Disadvantage: Effects of Small-Scale Schooling on Student Achievement in Alaska. *Journal of Research in Rural Education* 9: 137-149.

Institute for Research and Reform in Education (January 2003). First things first: A framework for successful school reform. A White Paper prepared for the Ewing Marion Kauffman Foundation.

Jacobson, Linda. (February, 2001). Research: Sizing Up Small Classes. *Education Week, Vol.20, number 24.* Washington, D.C.: Editorial Projects in Education.

Jewell, R. S. (1989). School and school district size relationships: Costs, results, minorities, and private school enrollments. *Education and Urban Society*, 21 (2), 140-153.

Jewell, R.W. (1989). School and school district size relationships: Costs, results, minorities, and private school enrollments. *Education and Urban Society*, 21(2), 140-153.

Johnson, D., & Johnson, R. (1990). Learning together and alone. New York: Prentice Hall.

Johnson, Jean, Ann Duffett, Steve Farkas and Kathleen Collins (2002). Sizing things up, What parents, teachers and students think about large and small high schools. Public Agenda.

Keller, Bess. (2001). Smaller Schools in Shared Space Seen as Recipe for Success. *Education Week*, 21(2). Washington, D.C.: Editorial Projects in Education.

Kemple, J.J. (1997). Career Academies: Communities of Support for Students and Teachers: Further Evidence from a 10-site Evaluation: Executive Summary. New York, NY: Manpower Research Development Corporation.

Klonsky, M. (1995b). Challenges of high school restructuring: The view from Philadelphia. Unpublished manuscript. Chicago, IL: University of Illinois-Chicago.

Klonsky, M., and Ford, P. (1994). One urban solution: Small schools. Educational Leadership, 51(8), 64-7.

Larson, R.L. (1991, March). Small is Beautiful: Innovation from the Inside Out. *Phi Delta Kappan*, pp. 550-554.

Lee, V. E., & Smith, J. B. (1995). Effects of high school restructuring and size on early gains in achievement and engagement. *Sociology of Education*, 68 (4), 241-70.

Lee, V. E., & Smith, J. B. (1996). *High school size: Which works best, and for whom*? Paper presented at the Annual Meeting of the American Educational Research Association, New York, NY.

Lee, V. E., Smith, J. B., & Croninger, R. G. (1995). Understanding high school restructuring effects on the equitable distribution of learning mathematics and science. Madison, WI: Center on Organization and Restructuring of Schools.

Lee, V., and J. Smith. (1994). Effects of High School Restructuring and Size on Gains in Achievement and Engagement for Early Secondary School Students. Madison, WI: National Center on the Organization and Restructuring of Schools.

Lee, V., and J. Smith. (1995). Collective Responsibility for Learning and Its Effects on Gains in Achievement for Early Secondary School Students. Madison, WI: National Center on the Organization and Restructuring of Schools.

Lee, V.E. & Smith, J.B. (1994, Fall). High school restructuring and student achievement: A new study finds strong links. Issues in restructuring schools: Madison: Center on Organization and Restructuring of Schools, Issue report #7.

Lee, V.E., & Smith, J.B. (Fall 1997). High school size: Which works best and for whom? *Educational Evaluation and Policy Analysis*, 19(3), 205-227.

Lee, V.E., & Smith, J.B. (October, 1995). Effects of high school restructuring and size on early gains in achievement and engagement. *Sociology of Education*, 68(4), 241-270.

Lee, V.E., A.S. Bryk, and J.B. Smith. (1993). The Organization of Effective Secondary Schools. In Review of Research in Education. Washington, D.C.: American Educational Research Association.

Lee, V.E., and Burkam, D.T. (2001). Dropping Out of High School: The Role of School Organization and Structure. Paper presented at "Dropouts of America: How Severe is the Problem?" Cambridge, MA: Harvard Graduate School of Education.

Lee, V.E., and Smith, J.B. (2001). *Restructuring High Schools for Equity and Excellence: What Works.* New York, NY: Teachers College Press.

Lee, V.E., Smerdon, B.A., Alfeld-Liro, C., and Brown, S.L. (2000). Inside large and small high schools: Curriculum and social relations. *Educational Evaluation and Policy Analysis*, 22(2), 147-171.

Legters, N.E. (1999). Small Learning Communities Meet School-to-Work: Whole School Restructuring for Urban Comprehensive High Schools. Report #31. Baltimore, MD: CRESPAR.

Lindsay, P. (1982). The effect of high school size on student participation, satisfaction, and attendance. *Educational Evaluation and Policy Analysis*, 4(1), 57-65.

Lindsay, P. (1984, Spring). High school size, population, activities, and young adult social participation: Some enduring effects of schooling. *Educational Evaluation and Policy Analysis*. 6(1), 73-83.

Los Angeles Unified School District, District B and Public Works, Inc. (2003) We Did It Together: The Transformation of Sun Valley Middle School. Pasadena, CA: Public Works, Inc.

Major Gates Foundation grants to support small high schools (July 16, 2004) *Education Week*, Vol. 23, number 40, 28-29.

McClintock, R., de Zengotita, T., Chou, L., & Moretti, F. Risk and Renewal: First Annual Report, 1991-1992. New York: New Laboratory for Teaching and Learning. McMullan, B.J., and Muncey, D.E. (1991). School-Within-A-School Restructuring and Faculty Divisiveness: Examples from a Study of the Coalition of Essential Schools. Working Paper #6. Providence, RI: Brown University, School Ethnography Project.

McPartland, J.M., Jordan, W., Legters, N., and Balfanz, R. (1997, October). Finding safety in small numbers. *Educational Leadership*, 55, 14-17.

McQuillan, P.J., and Muncey, D.E. (1991). School-Within-a-School Restructuring and Faculty Divisiveness: Examples from a Study of the Coalition of Essential Schools. Working Paper #6. Providence, RI: Brown University, School Ethnography Project.

Meier, D. (1993, Fall). A talk on small schools. BPI Newsletter.

Meier, D. (1995). The power of their ideas: Lessons for America from a small school in Harlem. Boston: Beacon Press.

Meier, D. (1995, January). How our small schools could be. Phi Delta Kappan.

Meier, D. (1997, January 31). For freshmen, a false start: Perils of ninth grade prompt freshmen to try new approaches. *Washington Post*, p. A1, A10.

Meier, D.W. (1996). The big benefits of smallness. Educational Leadership, 54(1), 12-15.

Meier, Deborah W. Small Schools, Big Results. The American School Board Journal.

Meier, Deborah. (February 2002). "Just Let Us Be; The Genesis of a Small Public School". *Educational Leadership*. Alexandria, VA: Association for Supervision and Curriculum Development.

Meier, Deborah. (May 1998) "Can the Odds Be Changed?" In *Small Schools, Big Imaginations: A Creative Look at Urban Public Schools*, edited by M. Fine and J.I. Somerville. Chicago, IL: Cross City Campaign for Urban School Reform, 85-92 (ED 427 127).

Mohr, N. (2000). Small schools are not miniature large schools: Potential pitfalls and implications for leadership. In M. Klonsky, W. Ayer, and G. Lyons (Eds.). *A Simple Justice*, (pp. 139-158). New York, NY: Teachers College Press.

Monk, D.H. (1987). Secondary school size and curriculum comprehensiveness. *Economics of Education Review*, 6,(2) pp. 137-150.

Moore, D.R., and Davenport, S. (1990). School choice: The new improved sorting machine. In W.L. Boyd and H.J. Walberg (Eds.) *Choice in Education: Potential and Problems* (pp.187-223). Berkeley, CA: McCutchan.

MPR Associates (1999). *Key high school reform strategies: An overview of research findings.* Berkeley, CA: MPR Associates, Inc. (http://www.ed.gov/offices/OVAE/has/research.html#improve) (7/5/00).

Muncey, D. E., & McQuillan, P. J. (1991). School-within-a-school restructuring and faculty divisiveness: Examples from a study of the Coalition of Essential Schools. Working Paper #6. Providence, RI: School Ethnography Project, Brown University.

Nathan, Joe and Karen Febey (2001). *Smaller, Safer, Saner, Successful Schools*. Minneapolis, MN: Center for School Change, Humphrey Institute of the University of Minnesota.

National Association of Secondary School Principals (NASSP) (1996). Breaking Ranks: Changing an American Institution. Reaton, VA: NASSP.

National Commission on the High School Senior Year (2001). The Lost Opportunity of Senior Year: Finding a Better Way. Washington, D.C.: National Commission on the High School Senior Year.

Newmann, F. (1995, April). *Reinventing the high school: The coalition campus school project in New York City* (*Comments as discussant*). Presentation at the annual meeting of the American Educational Research Association, San Francisco, CA.

Newmann, F. (1996). Center on Organization and Restructuring Schools: Activities and Accomplishments, 1990-1996 (1996). Madison, WI: University of Wisconsin, School of Educationa, Wisconsin Center for Education Research.

Nickle, M.N., Flynt, F.C., Poynter, S.D., and Rees, J.A., Jr. (1990). Does it make a difference if you change the structure? School-within-a-school. *Phi Delta Kappan*, 72(2), 148-152.

Noguera, Pedro A. (2002). Beyond Size; The Challenge of High School Reform. *Educational Leadership*. Alexandria, VA: Association for Supervision and Curriculum Development.

North Central Regional Educational Laboratory (1994). Resilience research: How can it help city schools?

O'Driscoll, P. (2000). Documenting the Effectiveness of School-to-Career Strategies: What Have We Learned? Pasadena, CA: Public Works, Inc.

Oxley, D. (1994). Organizing schools into small units: Alternatives to homogeneous grouping. *Phi Delta Kappan*, 75 (7), 521-526.

Oxley, D. (1997). Theory and practice of school communities. *Educational Administration Quarterly*, 33 (suppl), 624-643.

Oxley, D., and J. McCabe. (1990). Restructuring Neighborhood High Schools: The House Plan Solution. New York: Public Education Association and Bank Street College of Education.

Page, L., C. Layzer, J. Schimmenti, L. Bernstein, and L. Horst (February 2002). *National evaluation of smaller learning communities, Literature review.* Prepared for US Department of Education, Planning and Evaluation Service, Washington, DC. Abt Associates, Inc.

Plath, K.R. (1965). Schools within schools: A study of high school organization. New York, NY: Teachers College, Bureau of Publications.

Polikoff, A. (1992, Fall). Why small schools work. BPI Newsletter.

Powell, A., Cohen, D., & Farrar, E. (1985). The shopping mall high school. New York: Houghton Mifflin.

Prasch, J., & Wampler, W.N. (1959, August). School within a school...A better way to organize a high school? *School Management*, 3, pp.33-36, 64-66.

Public Education Association. (1992). Small school's operating costs: Reversing assumptions about economies of scale. New York: The Association.

Ramierz, A. (1990). High school size and equality of educational opportunity. *Journal of Rural and Small Schools*, 4 (2), 12-19.

Ramsey, R.D., Henson, O.M., & Hula, H.L. (1967). *The schools-within-a-school program*. West Nyack, NY: Parker.

Raywid, M. A. (1995). *The subschools/small schools movement--taking stock*. Madison, WI: Center on Organization and Restructuring of Schools. (ERIC Document Reproduction Service No. ED 397 490)

Raywid, M. A. (1996a). Taking stock: The movement to create mini-schools, schools-within-schools, and separate small schools. Urban Diversity Series No. 108. New York: ERIC Clearinghouse on Urban Education. Madison, WI: Center on Organization and Restructuring of Schools. (ERIC Document Reproduction Service No. ED 396 045)

Raywid, M. A. (1996b). The Wadleigh complex: A dream that soured. In W. Boyd, R. Crowson, & H. Mawhinney (Eds.), *The politics of education and the new institutionalism: Reinventing the American school.* Philadelphia: Falmer.

Raywid, M.A. & Henderson, H. (1994). 'Small' revolution in New York City. *Journal of Negro Education*, 63 (1).

Raywid, M.A. (1994). A school that really works: Urban academy. *Journal of Negro Education*, 63(1), 93-110.

Raywid, M.A. (1997). Small schools: A reform that works. Educational Leadership, 55(4), 34-38.

Raywid, Mary Ann (1996) Taking Stock: The Movement to Create Mini-Schools, Schools-Within Schools, and Separate Small Schools. New York: ERIC Clearinghouse on Urban Education, Teachers College, Columbia University.

Raywid, Mary Ann. "The Policy Environments of Small Schools and Schools-Within-Schools". *Educational Leadership*. Alexandria, VA: Association for Supervision and Curriculum Development, February 2002.

Robinson-Lewis, G. (1991). Summative Evaluation of the School-Within-a-school (SWAS) Program: 1988-1989, 1989-1990, 1990-1991. Kansas City, MO: Kansas City School District, 1991. (ERIC Document Reproduction Service no. ED346 203).

Rumberger, R.W., and Thomas, S.L. (2000). The distribution of dropout and turnover rates among urban and suburban high schools. *Sociology of Education*, 73(1), 39-67.

Sack, Joetta L. (February 2002). "Smaller Classes Under Scrutiny in Calif. Schools". *Education Week, Vol.21, number 24.* Washington, D.C.: Editorial Projects in Education.

Shireman, R. (May 12, 2004) "Where do 'multiple pathways' take us?" *Education Week*, Vol. 23, number 36.

Sizer, T. (1985). Horace's compromise: The dilemma of the American High School. Boston: Houghton Mifflin.

Sizer, T.R. (1986). Rebuilding: First steps by the Coalition of Essential Schools. *Phi Delta Kappan*, 68(1), 38-42.

Sizer, T.R. (1999). No two are quite alike. Educational Leadership, 57(1), 6-11.

Small Schools Project (Summer 2001a). "About Small Schools." Seattle, WA: Small Schools Project, Center on Reinventing Public Education, University of Washington.

Snyder, D. (1997). 4-Block Scheduling: A Case Study of Data Analysis of One High School After Two Years. Paper presented at the Annual Meeting of the Midwest Educational Research Association, Chicago, IL. (ERIC Document Reproduction Service no. ED 414 626).

Spencer, W.A., and Lowe, C. (1994). *The use of Block Period for Instruction: A Report and Evaluation*. Paper presented at the Annual Conference of the Mid-South Educational Research Association, Nashville, TN. (ERIC Document Reproduction Service no. ED 387-941).

Stanley, A. and Gifford, L.J. (1998). *The Feasibility of 4x4 Block Scheduling in Secondary Schools: A Review of the Literature*. Paper presented at the Annual Conference of the Mid-South Educational Research Association, New Orleans, LA. (ERIC Document Reproduction Service no. ED 429 333).

Steinberg, Adria and Lili Allen. (2000). *From Large to Small: Strategies for Personalizing the High School.* Washington, D.C.: Office of Educational Research and Improvement (OERI), U.S. Department of Education.

Stern, D., Dayton, C, Paik, I., and Weisberg, A. (1989). Benefits and costs of dropout preventing in a high school program combining academic and vocational education: Third year results from replications of the California Penninsula Academies *Educational Evaluation and Policy Analysis*, 11(4), 405-416.

Stern, D., M. Raby, and C. Dayton. (1992). Career Academies: Partnerships for Reconstructing American High Schools. San Francisco: Jossey-Bass.

Stevens, N. G., & Peltier, G. L. (1994). A review of research on small-school student participation in extracurricular activities. *Journal of Research in Rural Education*, 10 (2), 116-120.

Templeton, I. (1972). School size. Educational Management Review, 13.

The North Hollywood School Family Annenberg Challenge. (2001) Los Angeles Unified School District, Los Angeles, CA.

Unite LA School 2 Career Evaluation: 1998-99 Final Report (1999). Public Works, Inc., Pasadena, CA.

United States Department of Education (1999). "Promising Results, Continuing Challenges: Final Report of the National Assessment of Title I", Washington, D.C.: United States Department of Education.

Vander Ark, T. (February 2002). "The Case for Small High Schools". *Educational Leadership*. Alexandria, VA: Association for Supervision and Curriculum Development, February 2002)

Vander Ark, T. (February 2002). "Personalization: Making Every School a Small School," *Principal Leadership.* 2 (6). High School edition.

Viadero, D. (February 2001) "Research: Smaller is Better". *Education Week, Vol.21, number 13.* Washington, D.C.: Editorial Projects in Education.

Viadero, D. (2001). "Changing times: Despite its popularity, block scheduling's effect on learning remains unproven." *Education Week*, 21(5), 38-40.

Viadero, D. (June 16, 2004) "Personal Touches." Education Week, Vol. 23, number 40.

Visher, M.G.; P. Teitelbaum.; and D. Emanuel. "Create Small Learning Environments Enabling Students and Teacher to Work Together." Key High School Reform Strategies: An Overview of Research Findings. New American High Schools: High School at the Leading Edge of Reform. Washington, D.C.: Office of Vocational and Adult Education, March 1999, 19-26 (ED 430 271). (http://ericae.net/ericdc/ED430271.htm) (9/30/03)

Walberg, H.J. (1992). On local control: is bigger better? In *Source Book on School and District Size, Cost and Quality*. Minneapolis, MN: Minnesota University, Hubert H. Humphrey Institute of Public Affairs. (ERIC Document Reproduction Service no. ED 361 164).

Wallach, C. and Lear, R. L. (March 2003). "An Early Report on Comprehensive High School Conversions." Seattle, WA: Small Schools Project, Center on Reinventing Public Education, University of Washington.

Wang, M.C., Reynolds, M.C., and Walberg, H.J. (December 1993/January 1994). Serving students at the margin. *Educational Leadership*, 52 (4), p. 15.

Warren, Eileen (1998). Four-year report on the effectiveness of California Partnership Academies 1992-92 – 1995-96. Sonoma State University, California Institute on Human Services for the California Department of Education.

Wasley, Patricia A. "Class Size, School Size, Small Classes, Small Schools: The Time is Now", *Educational Leadership, Vol. 59, Number 5*. Alexandria, VA: Association for Supervision and Curriculum Development, February 2002.

Wehlage, G., Rutter, R., Smith, G., Lesko, N., & Fernandez, R. (1989). *Reducing the risk: Schools as communities of support*. Philadelphia, PA: Falmer.

Wehlage, G.G., R.A. Rutter, and A. Tumbaugh. (March 1987). "A Program Model for At-Risk High School Students." Educational Leadership 45: 70-73.

Weissmann, D. (1992, December). Reform heavy weights promote small schools. *Catalyst: Voices of Chicago School Reform*.

Wood, G. (1993). Schools that work: America's most innovative public education programs. New York: Plume.

Appendix B: Survey Results

2006-07 Small Learning Communities Confidential Student Survey (N = 211)



District:INGLEWOOD USDCohort:IVSchool:MORNINGSIDE HIGH SCHOOLGrade:9th

ecti	on I: Experiences in Your Classes	Strongly Disagree	Disagree	Agree	Strongly Agree
1	My classes are interesting and challenging.	5%	36%	54%	5%
2	I have the opportunity to do assignments and projects about interesting topics in class.	4%	35%	54%	6%
3	Teachers and administrators encourage me to challenge myself.	8%	22%	48%	23%
4	My teachers know my name and the names of my friends in class.	3%	16%	48%	33%
5	I have been encouraged to take AP and advanced classes.	23%	44%	30%	3%
6	I can get tutoring and other help if I'm having trouble in school.	6%	12%	50%	32%
7	My teachers are clear about what they expect from me.	4%	21%	57%	17%
8	My teachers are fair about how they grade me.	17%	25%	44%	14%
9	Teachers teach academic subject matter at a high level.	11%	28%	55%	6%
10	My parents feel comfortable with my teachers if they have questions or need information.	10%	17%	51%	22%
11	I have an adult at this school that I can go to for help with school and for personal support.	14%	24%	34%	28%
12	The classes I take incorporate my life experiences and my culture.	20%	34%	39%	6%
13	My classes have encouraged me to consider further education after high school.	9%	20%	45%	26%
14	I have worked with a counselor to develop a written educational plan that reflects my needs and interests.	25%	47%	21%	7%
15	I have worked with a teacher to develop a written educational plan that reflects my needs and interests.	28%	40%	26%	6%
16	I will be prepared to enter college when I am finished with high school.	6%	13%	42%	39%
17	I feel safe when I am at school.	27%	34%	35%	3%
18	I talk to my teachers or a counselor regularly about my high school educational plan.	27%	49%	20%	3%
19	I feel that I belong to a school-wide community.	16%	36%	45%	4%
20	The classes I take relate to my future college and career goals.	13%	32%	41%	13%
21	I will be prepared for employment when I am finished with high school.	6%	12%	49%	33%
22	I have the support I need at home to complete my homework and do well in school.	5%	9%	41%	45%

Section II: High School Learning Experiences

- 23 Are you currently involved in any of the following program offerings at this school? (mark all that apply)
 - 8% An Academy (a program made up of a group of students and teachers who share classes usually organized around a career theme)
 - 5% A Pathway (where students take a sequence of courses that lead them to their future college and career goals)
 - 9% A ROP class or program (such as business, culinary arts, or drafting)
 - 9% A House (where groups of students are assigned to a set of teachers who help them figure out classes, what they want to do after high school)
 - 7% A Magnet program (with a specialty core focus such as math, science, creative arts, or a career theme or cluster
 - 53% An Advisory program (where groups of students are assigned to a teacher or other faculty member on a regular basis to support planning my school and career decisions)
 - 16% None
 - 21% I don't know
 - 4% Other
- 24 Has anyone at the school told you about the kinds of programs listed in Question #23 that are available to you at this school?
 - 37% Yes
 - 33% No
 - 29% Don't Know
- 25 Are you assigned to a teacher or other staff member to help you plan your education in high school and after you graduate?
 - 17% Yes
 - 58% No
 - 25% Don't Know
- 26 How many times have you met with a counselor this school year?
 - 17% None
 - 33% 1-2 times
 - 23% 3-5 times
 - 26% more than 5 times
- 27 If you met with a counselor this school year, please select the reason or reasons you met. (mark all that apply)
 - 56% Selecting courses
 - 35% Help with a personal issue
 - 13% Planning for college

28 This school year, have you participated in any of the following activities? (mark all that apply)

- 33% After-school program
- 4% Internship
- 0% Community service project
- 1% Career/interest inventory
- 2% College fair
- 3% Guest speakers in your class

- 2% College class
- 2% Work experience
- 5% Job shadowing
- 43% Career fair
- 24% Field trip

29 What ADULT at this school is MOST helpful to you in planning for high school and life after high school? (mark all that apply)

- 52% Teacher
- 6% Principal
- 2% Assistant Principal
- 3% Office staff member
- 19% Coach

- 30% Counselor
- 2% Career center staff
- 1% Library staff member
- 2% Teaching assistant
- 9% Someone else at the school (what is their job)

Section III: About You

30 What grade are you in?

- 100% 9th
- 0% 10th
- 0% 11th
- 0% 12th

31 Are you:

- 51% Male
- 49% Female

32 What is your ethnicity?

- 30% African American
- 1% American Indian or Alaskan Native
- 1% Asian American/Pacific Islander
- 62% Hispanic/Latino
- 1% White/Caucasian
- 4% Other

33 Have you taken or are you currently taking an AP class?

- 8% Yes
- 92% No

34 What is the highest-level math class that you have taken, including any class that you are currently taking?

- 2% No math
- 81% Algebra I
- 12% Geometry
- 4% Algebra II
- 0% Trigonometry
- 0% Calculus
- 0% Other

35 What are your plans after high school graduation? (mark all that apply)

- 0% Attend a trade or vocational school
- 12% Attend a two-year college
- 51% Attend a four-year college or university
- 22% Find a full-time job
- 25% Find a part-time job
- 4% Join the military
- 1% Become an apprentice
- 13% Other
- 14% Don't know

36 School is:

- 15% Easy
- 66% Just right
- 19% Hard

37 I am a student in (if your school has tracks):

- N/A Track A
- N/A Track B
- N/A Track C

2006-07 Small Learning Communities Confidential Student Survey (N = 168)



District:INGLEWOOD USDCohort:IVSchool:MORNINGSIDE HIGH SCHOOLGrade:12th

ecti	on I: Experiences in Your Classes	Strongly Disagree	Disagree	Agree	Strongly Agree
1	My classes are interesting and challenging.	2%	35%	55%	7%
2	I have the opportunity to do assignments and projects about interesting topics in class.	2%	28%	64%	6%
3	Teachers and administrators encourage me to challenge myself.	4%	27%	55%	14%
4	My teachers know my name and the names of my friends in class.	1%	10%	55%	34%
5	I have been encouraged to take AP and advanced classes.	17%	36%	34%	13%
6	I can get tutoring and other help if I'm having trouble in school.	7%	23%	58%	11%
7	My teachers are clear about what they expect from me.	4%	20%	57%	20%
8	My teachers are fair about how they grade me.	5%	17%	64%	14%
9	Teachers teach academic subject matter at a high level.	4%	26%	63%	7%
10	My parents feel comfortable with my teachers if they have questions or need information.	5%	22%	59%	14%
11	I have an adult at this school that I can go to for help with school and for personal support.	10%	23%	46%	21%
12	The classes I take incorporate my life experiences and my culture.	9%	41%	45%	5%
13	My classes have encouraged me to consider further education after high school.	7%	12%	55%	27%
14	I have worked with a counselor to develop a written educational plan that reflects my needs and interests.	13%	41%	35%	11%
15	I have worked with a teacher to develop a written educational plan that reflects my needs and interests.	12%	46%	35%	7%
16	I will be prepared to enter college when I am finished with high school.	4%	15%	58%	23%
17	I feel safe when I am at school.	18%	34%	41%	7%
18	I talk to my teachers or a counselor regularly about my high school educational plan.	6%	38%	48%	8%
19	I feel that I belong to a school-wide community.	7%	33%	55%	5%
20	The classes I take relate to my future college and career goals.	12%	30%	48%	10%
21	I will be prepared for employment when I am finished with high school.	3%	21%	54%	22%
22	I have the support I need at home to complete my homework and do well in school.	4%	8%	51%	36%

Section II: High School Learning Experiences

- 23 Are you currently involved in any of the following program offerings at this school? (mark all that apply)
 - 5% An Academy (a program made up of a group of students and teachers who share classes usually organized around a career theme)
 - 9% A Pathway (where students take a sequence of courses that lead them to their future college and career goals)
 - 12% A ROP class or program (such as business, culinary arts, or drafting)
 - 2% A House (where groups of students are assigned to a set of teachers who help them figure out classes, what they want to do after high school)
 - 4% A Magnet program (with a specialty core focus such as math, science, creative arts, or a career theme or cluster
 - 67% An Advisory program (where groups of students are assigned to a teacher or other faculty member on a regular basis to support planning my school and career decisions)
 - 21% None
 - 4% I don't know
 - 4% Other
- 24 Has anyone at the school told you about the kinds of programs listed in Question #23 that are available to you at this school?
 - 39% Yes
 - 46% No
 - 14% Don't Know
- 25 Are you assigned to a teacher or other staff member to help you plan your education in high school and after you graduate?
 - 50% Yes
 - 40% No
 - 10% Don't Know
- 26 How many times have you met with a counselor this school year?
 - 1% None
 - 12% 1-2 times
 - 26% 3-5 times
 - 61% more than 5 times

27 If you met with a counselor this school year, please select the reason or reasons you met. (mark all that apply)

- 73% Selecting courses
- 26% Help with a personal issue
- 49% Planning for college

28 This school year, have you participated in any of the following activities? (mark all that apply)

- 21% After-school program
- 18% Internship
- 7% Community service project
- 18% Career/interest inventory
- 11% College fair
- 9% Guest speakers in your class
- 29 What ADULT at this school is MOST helpful to you in planning for high school and life after high school? (mark all that apply)
 - 48% Teacher
 - 3% Principal
 - 10% Assistant Principal
 - 2% Office staff member
 - 13% Coach

53% Counselor

7%

6%

29% Career center staff

College class

20% Job shadowing

55% Career fair

63% Field trip

Work experience

- 2% Library staff member
- 1% Teaching assistant
- 4% Someone else at the school (what is their job)

Section III: About You

30 What grade are you in?

- 0% 9th
- 0% 10th
- 0% 11th
- 100% 12th

31 Are you:

- 48% Male
- 52% Female

32 What is your ethnicity?

- 32% African American
- 1% American Indian or Alaskan Native
- 1% Asian American/Pacific Islander
- 63% Hispanic/Latino
- 0% White/Caucasian
- 4% Other

33 Have you taken or are you currently taking an AP class?

- 43% Yes
- 57% No

34 What is the highest-level math class that you have taken, including any class that you are currently taking?

- 0% No math
- 2% Algebra I
- 1% Geometry
- 57% Algebra II
- 3% Trigonometry
- 13% Calculus
- 25% Other

35 What are your plans after high school graduation? (mark all that apply)

- 8% Attend a trade or vocational school
- 40% Attend a two-year college
- 51% Attend a four-year college or university
- 13% Find a full-time job
- 33% Find a part-time job
- 2% Join the military
- 1% Become an apprentice
- 7% Other
- 2% Don't know

36 School is:

- 16% Easy
- 75% Just right
- 8% Hard

37 I am a student in (if your school has tracks):

- N/A Track A
- N/A Track B
- N/A Track C

2006-07 Small Learning Communities

S	Staff Survey (N = 44)						
Di Co So	District: INGLEWOOD USD Cohort: IV School: MORNINGSIDE HIGH SCHOO						
Re	espond	ent Characteristics					
1	Stake	nolder Group	2.	Years	at School		
	9% 80% 0% 11% 0%	Administration Classroom Teacher Teaching Assistant Counselor Other Classified		32% 27% 16% 25%	2 years or less 3-5 6-10 More than 10		
4	Subjec (shade	t (Teachers Only) all that apply)	5.	Grade (Teach (shade	Levels Currently ers Only) all that apply)		
	14% 14% 14% 16% 5%	Social Studies Math Science Special Ed		48% 48% 48% 48%	9th 10th 11th 12th		

- 9% Career Technical education/ROP
- 16% Other

The Small Learning Communities initiative is designed to personalize and scale down the educational experiences of students in large, comprehensive high schools. Small learning communities are structured in a variety of ways but typically consist of a group of students (between 100 and 500 students) who have the opportunity to develop personal relationships with small groups of peers and teachers in separate and distinct units within the school.

Teaching

I am currently assigned to work in a Small Learning Community at this school: 6

52% Yes

48% No

If you answered Yes above, which of the following Small Learning Communities are you assigned to (shade all that apply):

- 9% An Academy (school-within-a-school organized around a theme)
- 0% Pathway (where students take a sequence of courses that lead them to their future college and career goals)
- 0% ROP course sequence or program (3 or more courses within a career technical sequence)
- 43% House (where students are divided into groups of several hundred, either across grade levels or by grade level to personalize the educational experience; for example, a freshman house)
- 0% Magnet program (with a specialty core focus such as math, science, creative arts, or a career theme or cluster
- 57% Advisory (where small groups of students are assigned to a faculty member and meet on a regular basis to provide opportunities to personalize education, support career and college planning, and check in on academic progress)
- 0% Other



3. Years Teaching (Teachers Only)

8%	2 years or less
24%	3-5
18%	6-10
50%	More than 10

IUSD Smaller Learning Communities Evaluation Report, 2006-07

Rigor	ous, Relevant Curriculum & Instruction	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
7	Students understand classroom academic expectations (i.e., they understand what standard they are being held accountable for).	5%	7%	77%	9%	2%
8	Instruction is culturally responsive and accommodates diverse student interests, learning styles and educational needs.	0%	23%	60%	14%	2%
9	School-wide instructional decisions usually take into account the needs of English Language Learner (ELL) students.	2%	14%	56%	16%	12%
10	Curriculum and instruction is organized so that all students are expected to learn and perform at high levels.	5%	12%	70%	7%	7%
11	There is a clear, connected and comprehensive model for monitoring student progress.	2%	35%	47%	5%	12%
12	Examination of disaggregated student data is a regular part of school planning and assessment.	2%	24%	60%	10%	5%
Teach	ning & Learning Teams	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
13	Small learning community team members meet regularly for planning, curriculum and activities.	14%	28%	26%	12%	21%
14	There is sufficient time for teachers to discuss and analyze student work in small learning community team meetings.	19%	33%	19%	0%	30%
15	There is sufficient time for teachers to support students' academic and personal needs and to help them plan for the future.	14%	40%	30%	2%	14%
16	Teachers are part of a professional community of practice that is collaborative and public.	10%	19%	50%	5%	17%
17	Professional development for the SLC initiative is designed by teachers and is specific for our school.	7%	42%	23%	2%	26%
18	Professional development promotes greater alignment of instruction with academic standards and accountability requirements.	5%	26%	44%	9%	16%
19	Small learning community topics are a regular feature of school-wide professional development.	7%	37%	26%	7%	23%
Perso	nalization	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
20	Students experience personalized instruction that is based on diverse learning styles and multiple intelligences.	5%	19%	63%	5%	9%
21	Students experience personalized instruction that blends academic rigor with projects that reflect students' interests, life experiences and culture.	2%	19%	60%	5%	14%
22	Students complete a written educational plan that encompasses goals for high school and postsecondary education with teachers and/or counselors.	21%	33%	16%	7%	23%
23	Students receive verbal counseling regarding their secondary and postsecondary course plan from teachers and/or counselors.	14%	7%	56%	12%	12%
24	Students receive career planning and guidance in the form of career inventories and assessments, job shadowing opportunities, field trips and career fairs.	7%	12%	63%	5%	12%

IUSD Smaller Learning Communities Evaluation Report, 2006-07

		Strongly	Diagarag	Aaroo	Strongly	Don't
Perso	nalization	Disagree	Disagree	Agree	Agree	Know
25	All students at this school have an adult advocating for their academic and personal needs.	12%	24%	29%	7%	29%
26	Students have opportunities to work with one or more teachers over multiple years (e.g., "looping" and "student advisories").	5%	16%	56%	12%	12%
27	Student discipline is not a major problem area at this school.	44%	37%	12%	2%	5%
28	Students experience a safe learning environment.	5%	19%	67%	5%	5%
29	Students have opportunities for learning that extend beyond the instructional day including after-school programs, college courses, internships, etc.	0%	21%	60%	14%	5%
30	There is a clear process for referring a student for academic intervention.	7%	35%	37%	14%	7%
ldenti	ty of Small Learning Communities	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
31	Small learning communities at this school have an educational philosophy that is shared by students, staff, families and community partners.	7%	32%	29%	2%	29%
32	Small learning communities have unique academic identities.	12%	26%	35%	2%	26%
33	Small learning communities have distinct physical boundaries.	10%	29%	38%	5%	19%
34	Small learning communities make decisions regarding curriculum, instruction and assessment.	7%	40%	26%	2%	26%
35	Small learning communities make decisions regarding budget, personnel and facilities.	16%	40%	7%	2%	35%
36	Small learning communities make decisions related to the master schedule and student programming.	12%	30%	26%	5%	28%
37	Small learning communities make decisions related to student conduct and issues of community safety.	12%	42%	16%	5%	26%
38	Small learning communities have administrators or teacher-directors who lead a cohesive faculty.	14%	24%	38%	7%	17%
39	The school's master schedule supports small learning communities.	2%	17%	46%	15%	20%
Inclus	sive Programs and Instructional Practices	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
40	Admission to small learning communities is open and inclusive.	0%	14%	42%	21%	23%
41	Small learning communities include heterogeneous groupings of students and are not tracked by student ability.	2%	5%	47%	19%	28%
42	Small learning communities provide information and outreach about their programs to <i>high school</i> students and parents.	2%	23%	37%	7%	30%

IUSD Smaller Learning Communities Evaluation Report, 2006-07

Inclus	sive Programs and Instructional Practices	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
43	Small learning communities provide information and outreach about their programs to <i>middle school</i> students and parents.	5%	16%	23%	5%	51%
44	Most staff at this school are committed to the principle that "all children can learn."	2%	12%	49%	21%	16%
Visio	n, Leadership & Management	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
45	The vision and goals for implementing small learning communities are well understood by staff.	14%	33%	30%	14%	9%
46	All staff members have a say in school decisions.	16%	47%	26%	2%	9%
47	The results of major school decisions are communicated to all staff.	17%	40%	33%	2%	7%
48	Most staff members at this school trust one another.	26%	26%	21%	2%	24%
49	This school has a strong leadership team that guides instruction and the implementation of the small learning communities initiative.	19%	33%	35%	7%	7%
50	The architectural design and/or use of space at this school support the implementation of small learning communities.	14%	21%	40%	14%	12%
Paren	t and Community Engagement	Strongly Disagree	Disagree	Agree	Strongly Agree	Don't Know
51	Parent and Community Engagement	12%	23%	40%	7%	19%
52	Community partners, employers and businesses are involved in the development of small learning communities.	14%	35%	12%	7%	33%
53	Parents are considered key collaborators and contributing members to the school community.	12%	23%	37%	7%	21%

Barriers to Implementing Small Learning Communities

Directions: In order to help evaluate the implementation of small learning communities at your school, we would like you to shade the top <u>three</u> areas that you see as the biggest barriers to implementation of the school's plan that exist today.

54. Teaching to rigourous academic standards	16%	55. Collaboration among staff	32%
56. Adequate professional development	30%	57. Adequacy of facilities	20%
58. Serving the needs of specific populations	2%	59. Meeting state accountability measures	14%
60. Academic support and intervention for students	27%	61. Parent/Community involvement	43%
62. School governance and decision-making	16%	63. Curricular access & equity	0%
64. School leadership and vision	23%	65. Student guidance & counseling	14%
66. Teacher teaming	9%	67. Resistance to change	30%
68. Adapting master schedule to SLCs	16%		
69. Other (please specify)	2%		

70. What is your opinion of the SLC initiative and your school's progress in implementation?

Thanks again for you participation. Questions regarding the survey should be directed to:

90 North Daisy Ave. Pasadena, CA 91107 phone # 626-564-9890

Appendix C: Inventory

	-
Rat Usi 1. 2. 3. 2. 3. 5. 4. 5. 5. 5.	
 ting Scale <i>ing rubric of effectiveness of implementation and coverage of school community</i> No Evidence of Implementation. Strategies have not been developed; few or no school community members involved and/or impacted; planning to take place in the future. Planning for Implementation. Strategies are in the planning stages; some or a few school community members are involved in planning; few or no school community members are being recruited for implementation and participation; some school community members are being recruited for implementation and participation; some school community members impacted. Developmental Implementation. Strategies have moved into implementation; implementation at the early developmental stages; impact on school community is growing. Solid Implementation. Strategies are in solid implemented; 100% of target school community is 	 Research Questions/Focus of Evaluation: Family and Community Connections: Engaged and involved parents, business and community members Improved school climate and safety Authentic Curriculum: Modification of instruction and the delivery of curriculum Personalization: Personalized learning environment for students Improved Student Outcomes and Achievement: Improved student achievement, eligibility and preparation for career and postsecondary ed, an and employment Structures for Collaboration and Teaching and Learning Teams: Support for teacher collaboration in the development of authentic curriculum Collaboration among staff to increase presonalization
3	L ork w
Ibric Areas: Vision, Leadership & Management Teaching and Learning Teams Rigorous, Relevant Curriculum & Instruction Inclusive Programs and Instructional Practices (SLC Structure) Accountability and Continuous Program Improvement School/District Support for SLCs Personalization Parent and Community Engagement	ent enrollment in postsecondary options ith small groups of students 75

participating and impact is positive.

School community includes students, teachers, staff, administrators, havents and community hartners as

s Visited: e attach site visit agenda and who was interviewed.
e attach site visit agenda and who was interviewed.
ription of school and overall SLC implementation strategies:
es of SLCs/Grade Level Configurations:
Strategies/accomplishments:
t to improve/in need of help:

5	
2	2
7	5
2	Ĕ.
5	0
ž	H
0	.
3	9
5	H
2	۲.
Ξ.	Ω
0	a
2	0
0.	5
Ś	1
0	6
2	Ť
Ð	Н.
3	<u> </u>
2	0
2	
é.	2
7	
2.	Ο.
ö	_
R	
~	\leq
5	5
2	2
2	
5	2
5	oin
2	2
2	<u> </u>
ã	R
-	
Ä	2
ς.	
2.	1
ž	<u> </u>
~	
2	õ
2	H
S	2
4	0
0	Б
2	
6	
ج	ند
Ø	R
~	÷
2	
2	

Implementation is characterized by a shared vision created by a group of educators, support staff, administrators, parents, and community who comprise the school learning community who assume responsibility for the learning of every student through a distinctive and focused standards-based curriculum.

	mission
	support the school's SLC vision and
	(8) Architectural design and uses of space
	the implementation of SLCs.
	school-wide improvement goals including
	(7) The overall school budget reflects
	leadership for the SLC initiative.
	demonstrate strong, engaged and positive
	(6) The principal and administrators
	clear to stakeholders.
	Roles for the implementation of SLCs are
	communicated to all staff and stakeholders.
	(5) Major decisions regarding SLCs are
	together.
	converting to SLCs and how each SLC fits
	(4) Stakeholders are aware of the vision for
	and implementation experience.
	reevaluated based on community input
	(3) The vision is periodically revisited and
	SLCs.
	to the execution of the school's vision for
	implementing and problem solving related
	(2) Stakeholders are involved in planning,
	Family and community connections
	Structures for collaborative teams
	Personalization
	Authentic & rigorous curriculum
	Improved student outcomes and achievement
	SLCs incorporates:
	(1) The vision related to implementing
Description of Strategies and Status of Implementation	INDICATORS Ratin (1-6)

1) What is working really well at this site in this area?

2) What needs the most improvement?

3) What technical assistance needs have been identified?

SLC teams have structured opportunities for interdisciplinary teaching and learning in order to collaborate and work with small groups of students. SLC teams share no more than 300 students and team members are assigned to SLCs for at least half of their school day. Common planning time and professional development resources are used to support SLC goals and to reflect on student work and performance and to adjust curriculum and instructional strategies.

INDICATORS	Rating (1-6)	Description of Strategies and Status of Implementation
(1) There is flexibility in scheduling that allows	Ì	
SLC teams to plan instruction and develop		
(2) ST C teams have common planning time and		
(2) SLC teams have common planning time and regular, ongoing meetings to discuss students, their		
work and to plan and implement SLC activities.		
(3) Teachers are part of a "professional community		
of practice" that is collaborative and public.		
(4) Professional development for the SLC initiative		
is designed by teachers and supports site-specific goals.		
(5) Professional development supports the use of		
student data and assessment results to inform		
instruction and to make mid-course corrections in		
(6) Professional development prepares teachers.		
counselors and other school staff to personalize the		
educational experience of students through the SLC initiative.		
(7) Professional development supports alignment		
of instruction with academic standards and		
accountability requirements.		
(8) Leadership development is included for SLC leads and administrators.		
(9) SLC teams share no more than 300 students in common for instruction.		
(10) SLC team members are assigned to SLCs for at least one half of their schedules.		
AVERAGE RATING:		

1) What is working really well at this site in this area?

2) What needs the most improvement?

3) What technical assistance needs have been identified?

A standards-based educational program embodies high expectations for every student to achieve grade-level standards and meet high school graduation requirements. Students are expected to meet college entrance requirements and are prepared for post-secondary education and training experiences and the world of work. Curriculum is adapted to individual student learning needs and is organized around topics of interest to students. Multiple forms of assessment provide opportunities to reflect personalized learning.

	- 5
INDICATORS	(1-6) Description of Strategies and Status of Implementation
(1) SLC course offerings align to district	
graduation and university admission	
(2) SLCs share clear expectations that	
align with state content and performance	
(3) Curriculum and instruction is	
organized according to individual SLC	
educational philosophy and may involve	
thematic, interdisciplinary units.	
(4) Curriculum and instruction is	
organized so that all students are expected	
to learn and perform at high levels.	
(5) Multiple forms of assessment reflect	
personalized learning and offer students	
opportunities to demonstrate learning.	
(6) Curriculum and instruction is	
articulated (up to post-secondary and	
down to middle schools) to provide a	
coherent educational experience resulting	
in students moving toward graduation.	
(7) Teachers adapt instruction based on	
the needs of individual students and attend	
to all learners including English language	
learners, standard English language	
learners, and students with special needs.	
(8) High quality, credentialed teachers	
teach in all SLCs.	
(9) Structured intervention is designed to	
meet individual student needs.	
(10) There is an adequate supply of basic	
classroom supplies, supplemental resources	
and Board adopted textbooks that are	
standards aligned.	
AVERAGE RATING:	

What is working really well at this site in this area?
 What needs the most improvement?

3) What technical assistance needs have been identified?

group of students and an administrator or teacher leader that leads a cohesive faculty team. SLC teams make decisions related to: curriculum, instruction in size from 100 to 300 students and students are blocked for at least one half of the school day in an SLC. and assessment; budget, personnel and facilities; master schedule and student programming; and student conduct and issues of community safety. SLCs range is based on students' and teachers' interest and choice to ensure equitable access. SLCs have a unique academic identity and include distinct, heterogeneous Each SLC has a coherent educational program and approach that is known and shared by students, staff, families and community partners. SLC membership

INDICATORS	Rating (1-6)	Description of Strategies and Status of Implementation
(1) SLCs have a coherent educational		
program and approach that is known and		
shared by students, staff, families and		
and tailored to diverse student needs.		
(2) Each SLC includes a distinct,		
heterogeneous group of students based on		
student interest and choice.		
(3) Each SLC has an administrator or lead		
teacher that leads a cohesive faculty team		
based on faculty interest and choice.		
collaborate with the teams		
(4) SLC teams make decisions related to:		
Curriculum, instruction, and		
assessment.		
Budget, personnel and facilities		
Master schedule and student		
Student conduct and issues of		
community safety		
(5) Access to SLCs is open and inclusive.		
SLCs are designed to be accessible to all		
(6) Size of SLCs are appropriate to the		
vision and mission (range in size from 100		
to 300 students).		
(7) Students within an SLC are together		
for at least 50% of their school day.		
AVERAGE RATING:		

3) What technical assistance needs have been identified?

2) What needs the most improvement?

1) What is working really well at this site in this area?
| | | (1) Ctaleabaldana |
|---|--|-------------------|
| Description of Strategies and Status of Implementation | Rating
(1-6) | INDICATORS |
| | | |
| nt aspects of the small learning community. | making decisions affecting the import | accountable for n |
| re used to make decisions. SLC teams retain primary responsibility, appropriate autonomy and are | al and external sources of school data | decisions. Intern |
| exercise leadership to ensure that student achievement and personal success is the intended result of all | LC work together, share expertise, an | Members of the S |
| Improvement Benchmark | ity and Continuous Program | Accountabili |

	•	
INDICATORS	Kating (1-6)	Description of Strategies and Status of Implementation
(1) Stakeholders display personal and		
collective responsibility for achieving the		
vision and mission for SLC vision and		
mission of success for all students.		
(2) Internal and external school data from		
multiple sources are used to make		
decisions.		
(3) Funds, time, personnel, partnerships		
and facilities are used to support the		
mission and vision of the school.		
(4) Decision-making and reporting		
processes incorporate the use of		
technology.		
(5) Student data is accessible by SLC.		
AVERAGE RATING:		

1) What is working really well at this site in this area?

2) What needs the most improvement?

3) What technical assistance needs have been identified?

School/District Support for SLCs Benchmark

School and district policies and practices support the implementation of SLCs. School-wide and departmental goals, professional development, scheduling and staffing align with and support SLC needs. Members of the SLC teams retain primary responsibility, appropriate autonomy and accountability for decisions related to individual SLCs.

INDICATORS	Rating (1-6)	Description of Strategies and Status of Implementation
(1) School-wide improvement goals align with SLC needs.		
(2) Department goals align with SLC needs		
(3) School and district professional		
development plans and resources accommodate SLC needs.		
(4) District policies support the		
implementations of SLCs including		
and SLC levels. District negotiates teacher		
union contract with provision to support SLC staffing needs.		
(5) School scheduling and staffing support the implementation of SLCs.		
(6) Options for Honors/AP classes are available across all programs.		
AVERAGE RATING:		

1) What is working really well at this site in this area?

2) What needs the most improvement?

3) What technical assistance needs have been identified?

INDICATORS	Rating (1-6)	Description of Strategies and Status of Implementation
(1) Students are known and valued by their peers and staff and have access to adult		
mentors/advisors and role models.		
(2) Students experience personalized instruction		
cultures. Instruction is based on diverse learning		
styles and multiple intelligences.		
(3) SLC serves a population of 300-500 students		
with increased teacher-adult contact and community responsibility.		
(4) Students prepare a written secondary course		
plan and postsecondary plan with teachers		
(5) Verbal counseling from teachers and/or		
counselors is a regular part of student		
educational programming.		
(b) Students receive college and career planning and guidance in the form of career inventories		
and assessments; job shadowing opportunities;		
(7) A 1 1 - 1		
(7) Adduts have available, timely, and comprehensible student data for advisory and		
course planning.		
(8) Students have opportunities to work with		
one or more teachers for multiple years in caring, supportive relationships (differing models of		
advisory, mentoring, dropout prevention)		
(9) Adults conduct parent outreach and		
conferences on student's personal needs to support students.		
(10) Students have opportunities for learning		
that extend beyond the instructional day		
including after-school programs, college courses,		
Internships, etc.		
academic intervention and support services as		
needed.		

Public Works, Inc.

Personalization Benchmark

Each student's educational experience is characterized by sustained and mutually respectful personal relationships with students, faculty and administrators. Students are known by a group of educators who advise/advocate for them and work closely with the student and his or her family over time. The size of the SLC is appropriate to its vision and mission ranging in size from 100 to 300 students.

- 1) What is working really well at this site in this area?
- 2) What needs the most improvement?
- 3) What technical assistance needs have been identified?

All members of the SLC are viewed as critical allies and are included in the school community (i.e., students, teachers, support staff, parents, administrators, business and community partners). Ongoing partnerships are aimed at supporting continuous improvement of student achievement and student's personal success. Authentic engagement of school partners leads to sustained participation of partners in decision making and implementation of school efforts.

INDICATORS	Rating (1-6)	Description of Strategies and Status of Implementation
(1) School encourages partnerships with		
community members, employers,		
postsecondary institutions and others		
necessary to implement SLCs.		
(2) Community partners, employers and		
businesses are involved in the development		
of curriculum, activities and other		
components to support SLCs.		
(3) Parents are considered key		
collaborators and contributing members to		
the school community.		
(4) Opportunities are provided for people		
to gather easily at appropriate times and		
locations.		
(5) Parents are involved in decision-		
making for their students including SLC		
choice, curriculum planning, student		
activities and future plans.		
AVERAGE RATING:		

1) What is working really well at this site in this area?

2) What needs the most improvement?

3) What technical assistance needs have been identified?